This is the peer reviewd version of the followng article:

Scaling entity resolution: A loosely schema-aware approach / Simonini, Giovanni; Gagliardelli, Luca; Bergamaschi, Sonia; Jagadish, H. V.. - In: INFORMATION SYSTEMS. - ISSN 0306-4379. - 83:(2019), pp. 145-165. [10.1016/j.is.2019.03.006]

Terms of use:

The terms and conditions for the reuse of this version of the manuscript are specified in the publishing policy. For all terms of use and more information see the publisher's website.

12/02/2025 06:26

Accepted Manuscript

Scaling entity resolution: A loosely schema-aware approach

Giovanni Simonini, Luca Gagliardelli, Sonia Bergamaschi, H.V. Jagadish

 PII:
 S0306-4379(18)30408-3

 DOI:
 https://doi.org/10.1016/j.is.2019.03.006

 Reference:
 IS 1380

To appear in: Information Systems

Received date : 3 August 2018 Revised date : 16 February 2019 Accepted date : 17 March 2019



Please cite this article as: G. Simonini, L. Gagliardelli, S. Bergamaschi et al., Scaling entity resolution: A loosely schema-aware approach, *Information Systems* (2019), https://doi.org/10.1016/j.is.2019.03.006

This is a PDF file of an unedited manuscript that has been accepted for publication. As a service to our customers we are providing this early version of the manuscript. The manuscript will undergo copyediting, typesetting, and review of the resulting proof before it is published in its final form. Please note that during the production process errors may be discovered which could affect the content, and all legal disclaimers that apply to the journal pertain.

Scaling Entity Resolution: A Loosely Schema aw re Approach

Giovanni Simonini^{a,1}, Luca Gagliardelli^b, Sonia Bergamasch[·], H √. Jagadish^c

^aMIT CSAIL ^bDepartment of Engineering "Enzo Ferrari", University of Moden and Res io Emilia, Italy ^cUniversity of Michigan, Ann Arbor, US

Abstract

In big data sources, real-world entities are typical, represented with a variety of schemata and formats (e.g., relational record. JSON objects, etc.). Different profiles (i.e., representations) of an $er^{\pm i + \cdots + r}$ contain redundant and/or inconsistent information. Thus identifying which profiles refer to the same entity is a fundamental task (called Entity Resolution) to unleash the value of big data. The naïve all-pairs comparison solution is impractical on large data, hence blocking methods are employed to partition a profile collection into (possibly overlapping) blocks and limit the unparisons to profiles that appear in the same block together. Meta-blocking is the task of restructuring a block collection, removing superfluous computisons. Existing meta-blocking approaches rely exclusively on schema-agnostic features, under the assumption that handling the schema variety of the lock of pay-off for such a task.

In this paper, we demonstrate now "loose" schema information (i.e., statistics collected directly from $u_{n} \circ dat_{n}$) can be exploited to enhance the quality of the blocks in a holistic *loosely schema-aware* (meta-)blocking approach that can be used to speed up y or flooring Entity Resolution algorithm. We call it *Blast* (Blocking with Loogely-rear Schema Techniques). We show how *Blast* can automatically extrement the loose schema information by adopting an LSH-based step for efficiently handing volume and schema heterogeneity of the data. Furthermore, we in the duce a novel meta-blocking algorithm that can be employed to efficiently elecute *Blast* on MapReduce-like systems (such as Apache Spark). Finally, we experimentally demonstrate, on real-world datasets, how *Blast* outperforms the state-or the-art (meta-)blocking approaches.

Keywords: 'ntit Resolution, Meta-blocking, Big Data Integration, Data Cleani^{*}.₅, Apac. - Spark 2010 MSC: UP01, 99-00

Email uddress: giovanni@csail.mit.edu (Giovanni Simonini)

Preprint submitted to Journal of $\square T_EX$ Templates

February 16, 2019

1. Introduction

In the context of big data, real-world entities are typically represented in a variety of formats, such as: records of relational databases, $R\Gamma$, riples, JSON objects, etc. Moreover, the *profiles* (i.e., the representation) of a real-world entity often contain overlapping, complementary and/or income tent information. Hence, a fundamental task for unleashing the value A this data is Entity Resolution (ER) [1, 2, 3], which aims to identify and recording the value profiles that refer to the same real-world entity.

Background: When the volume of the data is 1 rge chicking all possible profile pairs to find *matches* is not a practical solution die to its quadratic complexity. For this reason, typically, signatures (b_{k} -king keys) are extracted from the profiles and employed to index them into bloks [4]. Then, the all-pairs comparison is limited to profiles within a blok is significantly reducing the complexity of ER.

Traditional blocking techniques typical and priori schema knowledge to devise good blocking keys by combining attribute values; hence suffering from two well-known issues:

- 1. Given a known schema, selecting which ttributes to combine requires either domain experts or labeled data to 'r.'n a classification algorithm [5].
- 2. If two datasets have different solutions a, a schema-alignment must be executed before ER. Unfortunately, the data is typically highly heterogeneous, noisy (missing/inconsistent data), and large in volume of schemata, so that traditional schema-alignment to chniques are no longer applicable [6, 7]. For instance, Google Base Contains over 10k entity types that are described with 100k unique schemata; in Loc' a scenario, performing and maintaining a schema alignment is impractical [6].

To work around the sistematic schema-agnostic blocking has been proposed [7, 8]. This approach ext. ets blocking keys from the profiles by treating them as *bags-of-words*. For instance, **Token Blocking** [7] considers each token in a profile as a blocking key; in other words, each pair of profiles sharing at least one token (recard) as to the attribute in which it appears) is considered as a candidate match, as shown in the example of Figure 1(a-b). By placing each profil in nultiple blocks, schema-agnostic techniques on one hand reduce the likelihood of missing matches, on the other hand increase the likelihood of placing pon-n. to import the same blocks. This allows the achievement of high means (i.e., the percentage of detected matching profiles), but at the expense of provision (e., the ratio between detected matching profiles and executed compare.

To inprove the precision of schema-agnostic blocking, *meta-blocking* approaches have been proposed [8]. Meta-blocking is the task of restructuring a set c^{c} blocks to retain only the most promising comparisons. Meta-blocking replacements a block collection as a weighted graph, called *blocking graph*, where



Figure 1: (a) A collection of entity profiles from a data lake where data is stored in different formats. (b) A block collection produced with Token Blocking; notice that the tokens appearing only in one profile on a converte any comparison (i.e., any block). (c) The derived blocking graph and by effect of meta-blocking: dashed lines represent pruned edges, and red ones to supe quous comparisons not removed. In this toy example, the weight of each edge form of two profiles p_i and p_j is equal to the number of blocks in which p_i and p_i for occur—other weighting functions can be employed [8]. For instance, p_1 and p_2 where only the block "Abram", so the weight of the edge that connects them is 1. Then, the pruning is performed computing a local threshold for each profile (e.g., the average of its edges' weights) and keeping only the edges having a weight higher than the local threshold. For instance, the weights of p_1 edges are $\{1, 3, 4\}$ and their the prime is 2.7, so the edge that connects p_1 with p_2 can be discarded, since 1 < 2.7.

each entity profile is a n_{c} e ar d an edge exists between two nodes if the corresponding profiles a bear at reast in one block together. The edges are weighted to capture the like indeet of a match. An example of a blocking graph is shown is Figure 1(c), when the weight of an edge is equal to the number of co-occurrences of its adjacent prof es in the blocks². Then, an edge-pruning scheme is applied to retain only the most promising ones. The most accurate strategy to prune edges is to consider for each node all its adjacent edges, and retain only those having a reight higher than the local average (Figure 1(c)). At the end of the process, each pair of nodes connected by an edge forms a new block.

Our Approa h: We observe that existing meta-blocking techniques exclusively leverage schema agnostic features to restructure a block collection. Inspired by the *attribute-match induction* approaches [7, 9], our idea is to exploit schema

²Co-occurrence in blocks is employed for the sake of the example; more sophisticated we $_{2}$ · 1g functions can be employed (see Section 3.3).

information extracted directly from the data for enhancing the quality of the blocks. Moreover, we argue that a holistic approach combining meta-lacking and *loosely schema-aware* techniques should be attempted. Hence, we introduce our approach called *Blast* (<u>B</u>locking with <u>Loosely-Aware Schema Fechniques</u>). *Blast* can easily collect significant statistics (e.g. similarities and erroppes of the values in the attributes) that approximately describe the data sources schema. This *loose* schema information is efficiently extracted even from highly heterogeneous and voluminous datasets, thanks to a novel LSI -based pre-processing step that guarantees a low time requirement. Then, the low schema information is exploited during both the blocking and meta-lacking phases to produce high quality block collections.

To get an intuition of the benefits of loose s. bema information, consider the example in Figure 2. Say that, among the different data sources, only the attributes about person names have similar value to some extent. Blast clusters together these attributes, while the others (" $n_{ell} + eno_{ell}$, similar" to each other) are grouped in a unique general cluster. Thus, us an disambiguate the token "Abram" as person name from its other use (e.g., street name). Consequently, the block associated to the token "Abram" is divered into two new blocks (Figure 2(a)) affecting the blocking graph: the verges of the edges $e_{p_1-p_4}$ and $e_{p_2-p_3}$ both decrease (Figure 2(b)). Therefore, t. local thresholds for meta-blocking changes, and one further superfluous ever (p_{1-p_4}) is correctly removed in the pruning step (Figure 2(b)). The precise n increases, while the recall remains the same. Yet, one superfluous comparison in the same the quality of the blocking. The intuition is that some attributes are more informative than others and can



Figure 2: (a) ' he locking key "Abram" is disambiguated by employing the loose schema information, as a consequence, the profiles p_1 and p_4 share one less block than before a this means also that the edge e_{1-4} decreases its weight accordingly, from 3 to 2. (b) The effect on the new blocking graph weights and on the meta-blocking process w.r.t. Figure 1(c): one further edge is correctly removed (e_{1-4} , dashed red $\lim_{r \to 1} c$ compared to Figure 1(c). As a matter of fact, e_{1-4} is now pruned, since it has a weight =2) lower than the local threshold of p_1 (=2.3); while in Figure 1(c), the weight of $_{1-4}$ is 3, which is greater than the local threshold of p_1 (=2.7)—notice that if the methation of e_{1-4} varies, the threshold of p_1 also changes, since the latter depends on d former.

generate more significant blocking keys. Blast measures the information content of an attribute through the Shannon entropy [10]. Then, it derives an a_{2} regate entropy measure for each cluster of attributes. Finally, it uses these values as a multiplicative coefficient in the weighting function of the blocking regan. For our toy example, the aggregate entropies are listed in Figure 4(a) and the final blocking graph after the pruning phase is showed in Figure 3(a)³, where the superfluous edge $e_{p_2-p_3}$ has now been correctly removed.



Figure 3: (a) Attribute entropy information of d its effect (b) on the blocking graph pruning. In this toy example, the weighting function is: $w(p_i, p_j) = \sum_{k \in K_i \cap K_j} \mathcal{H}(b_k)$, where K_i and K_j are the set of blocking key of p_i and p_j respectively, and $\mathcal{H}(b_k)$ is the aggregate entropy of the chain to thich b_k belongs to. In (b), the effect on the new blocking graph weights and on the meta-blocking process is shown w.r.t. Figure 2(b): one further edge is correctly removed (e_{2-3} , dashed red line) compared to Figure 2(b). As a matter $e^{e_{1-2}}$, e_{2-3} is now pruned, since it has a weight (=6) lower than the local threshol of p_1 (=6.3).

Contributions: Over 1, we make the following main contributions:

- an approach to au^{*} oma. ^{*}all^{*} extract *loose schema information* from a dataset based on an attr^{*} oute-match induction technique;
- an unsupervised graph-based meta-blocking approach able to leverage this loose scheme information;
- an LSH-bised a 'ribute-match induction technique for efficiently scale to large datasets' wit¹ a high number of attributes;
- an algorith. tc efficiently run *Blast* (and any other graph-based meta-blocking met.iod) on MapReduce-like systems, to take full advantage of a parallel and dis ributed computation;

³ For t1 sake of the example the weights are computed starting from the blocking graph or \forall igure '(b); in the actual processing only one blocking graph is generated, and a unique pruning step is performed.

ACCEPTED MANUSCRIPT

• the evaluation of our approach on seven real-world datasets, s' owi'g how Blast outperforms the state-of-the-art meta-blocking methods.

A preliminary version of *Blast* was published in [11]. In this pap x, "last has been extended to take advantage of a parallel and distributed in putation for significantly reducing the overall execution time of the Fill process, which is typically onerous in the big data context. We propose bis adcast neta-blocking (Section 4): a novel algorithm to run any graph-based nota-blocking method (including *Blast*) on distributed MapReduce-like systems, such as Apache Spark. Finally, we provide more extensive experiments on lorge scale datasets⁴, which showcase that our solution efficiently scales on Mapreduce-like systems and outperforms the state-of-the-art meta-blocking method is (Section 5).

Organization: The remainder of this paper is structured as follows. Section 2 provides preliminaries. Section 3 preset is $D_{1,2}^{-1}$ and Section 4 describes basic concepts for distributed meta-blocking on AppReduce-like systems and discusses *Blast* parallelization. Section 5 presents the datasets, the evaluation metrics, and the experiments. Section 6 examines the related work. Finally, Section 7 concludes the paper.

2. Preliminaries

This section defines preparato. To the and notation employed throughout the paper.

2.1. Blocking for Entity Resolution

An entity profile is a while composed of a unique identifier and a set of name-value pairs $\langle a, v \rangle \land A_{\mathcal{P}}$ is the set of possible attributes a associated to a profile collection \mathcal{P} . In p of le collection \mathcal{P} is a set of profiles. Two profiles $p_i, p_j \in \mathcal{P}$ are matrix $p_i \approx p_j$ if they refer to the same real world object; Entity Resolution (CR) is the task of identifying those matches given \mathcal{P} .

The naive solution to ER implies $|\mathcal{P}_1| \cdot |\mathcal{P}_2|$ comparisons, where $|\mathcal{P}_i|$ is the cardinality of a profile collection \mathcal{P}_i . Blocking approaches aim to reduce this complexity by indefining similar profiles into blocks according to a blocking key (i.e., the index, or criterion), restricting the actual comparisons of profiles to those apper ring in the same block.

Given the rate et of Figure 1(a), an example of *schema-agnostic* blocking key is shown in F gure 1(b). Otherwise, a *schema-based* blocking key might be the value of the attribute "name"; meaning that only profiles that have the same value for "name" will be compared (the dataset in Figure 1(a) would require a schema plicement). A set of blocks \mathcal{B} is called *block collection*, and its *aggregate c* rainality is $\|\mathcal{B}\| = \sum_{b_i \in \mathcal{B}} \|b_i\|$, where $\|b_i\|$ is the number of comparisons implied live the block b_i .

^{&#}x27;Two additional datasets are introduced in Section 5: citation3 and freebase.

We follow best practices to establish the quality of a block colletion [7, 12]: the problem of determining if two profiles actually refer to the same real world object is the task of the *Entity Resolution Algorithm*. We assume that a is such an algorithm able to determine whether two profiles are mathing or not. In fact, *Blast* is independent of the *Entity Resolution Algorithm* employed, just as the other state-of-the-art blocking techniques compared in this poer [12, 13].

2.1.1. Dirty ER and Clean-Clean ER

Papadakis et al. [12] have formalized two types of ER to be Dirty ER and Clean-Clean ER. The former refers to those scenario, where ER is applied to a single data source containing duplicates; this problem is used a nown in literature as deduplication [14]. In the latter, ER is applied to two more data sources, which are considered "clean", i.e., each source considered singularly does not contain duplicate. This type of ER is also known as Record Linkage [14]. As in [11, 12, 13, 15, 16], in this work, we adopt the classification as well.

Notice that, in *Clean-Clean ER* the comparison profiles that belong to the same data source are avoided (for , w underlying blocking technique) [12]. Hence, the number of comparisons $||b_i|| \operatorname{req}$ ired for a block b_i depends on the type of ER: for *Dirty ER*, a block produce $||b_i|| = \binom{|b_i|}{2}$, where $|b_i|$ is the cardinality of the block and all the possible comparisons are considered; while, for *Clean-Clean ER*, a block produce $\| c_i \| = \sum_{j=1}^N \sum_{k=j+1}^N |b_i^j| \cdot |b_i^k|$, where b_i^j is the subset of \mathcal{P}^j indexed in the block b_i , and N is the number of data sources.

2.1.2. Metrics

We employ *Recall* and P^r is ion to evaluate the quality of a block collection \mathcal{B} , as in [1]. The recall reasure, the portion of duplicate profiles that are placed in at least one block, while the precision measures the portion of useful comparisons, i.e., those that dete t a match. Formally, precision and recall of a blocking method is defermined from the block collection \mathcal{B} that it generates:

$$e_{\mathcal{T}} ll = \frac{|\mathcal{L}^{\mathcal{S}}|}{|\mathcal{D}^{\mathcal{P}}|}; \qquad precision = \frac{|\mathcal{D}^{\mathcal{B}}|}{||\mathcal{B}||};$$

where $\mathcal{D}^{\mathcal{B}}$ is the set of duplicates appearing in \mathcal{B} and $\mathcal{D}^{\mathcal{P}}$ is the set of all duplicates in the value of \mathcal{P} .

Typically, chema-a_bnostic blocking yields high recall, but at the expense of precision The low precision is due to the unnecessary comparisons: redundant comparisons $\neg t_{\ell}$ the comparison of profiles more than once; and superfluous comparisons $\neg t_{\ell}$ is the comparison of non-matching profiles $(p_i \not\approx p_i)$.

For instance, considering the block collection of Figure 1(b), the pair of profiles (p_1, p_2) ppears in many blocks ("Car", "Main", etc.), thus, if all the blocks a e evalvated as traditional blocking techniques do [4] (i.e., without performing redundant comparisons. Figure 1(b) also provides examples of *superfluous* comparisons, be a the superfluence between p_1 and p_2 are compared more than once performing redundant comparisons. Figure 1(b) also provides examples of *superfluous* comparisons,

 \dots has the comparisons between p_2 and p_5 , and between p_4 and p_5 , entailed

by the block "Ellen"—we call these comparisons superfluous becau p_5 do not match neither with p_2 nor p_4 .

Attribute-match induction⁵ approaches can be employed to enhance schemaagnostic blocking by limiting the superfluous comparisons. Meta-blocking is the state-of-the-art approach to reduce both superfluous and reduindant comparisons from an existing block collection. In the following we formally as the attributematch induction and meta-blocking.

2.2. Attribute-match Induction

The goal of attribute-match induction is to ide tify \mathcal{J}_{1} ups of similar attributes between two profile collections \mathcal{P}_{1} and \mathcal{P}_{2} for the distribution of the attribute values, without exploiting the semantics of the distribute names. This information can be exploited to support a schema-agno. The blocking technique, i.e., to disambiguate blocking keys according to \mathcal{L}_{2} attribute group from which they are derived (e.g. tokens "Abram" in Figure 1(b)).

Definition 1. Attribute-match induction. Given two profile collections $\mathcal{P}_1, \mathcal{P}_2$, attribute-match induction is the task of identify. Fairs $\{\langle a_i, a_j \rangle \mid a_i \in A_{\mathcal{P}_1}, a_j \in A_{\mathcal{P}_2}\}$ of similar attributes according to a simple of y measure, and use those pairs to produce the attributes partitioning. i.e., f partition the attribute name space $(A_{\mathcal{P}_1} \times A_{\mathcal{P}_2})$ in non-overlapping clusing.

An attribute-match induction ask can be defined through four components, formalized in the followin. (i) the value transformation function (ii) the attribute representation m odel, (ii) the similarity measure to match attributes, and (iv) the clustering algo ithm.

- (i) The value transformation γ is the function. Given two profile collections \mathcal{P}_1 and \mathcal{P}_2 , each tribute is represented as a tuple $\langle a_j, \tau(V_{a_j}) \rangle$, where: $a_j \in A_{\mathcal{P}_i}$ is an attribute name; V_{a_j} is the set of values that an attribute a_j can assume in \mathcal{P}_i ; and τ is a value transformation function returning the set of transformed values $\{\tau(v) : v \in V_{a_j}\}$. The function τ generally is a concatent ion of text transformation functions (e.g. tokenization, stopword's removal, lemmatization). Given a τ transformation function, the set of v ssib's values in the profile collections is $T_A = T_{a_{\mathcal{P}_1}} \cap T_{a_{\mathcal{P}_2}}$, where $T = \bigcup_{a_i \in A_{\mathcal{P}}} \tau(V_{a_i})$.
- (ii) The attribute representation model. Each attribute a_i is represented as γ in ctor \mathcal{T}_i (called the *profile* of a_i), where each element $v_{in} \in \mathcal{T}_i$ is as ociated to an element $t_n \in T_A$. If $t_n \notin \tau(V_{a_i})$, then v_{in} is equal to zero.

⁵We call *attribute-match induction* the general approach to group similar attributes, while we reter to the specific technique proposed in [7] with *Attribute Clustering*.

While, if $t_n \in \tau(V_{a_i})$, then v_{in} assumes a value computed upplying a weighting function, such as [7]: TF- $IDF(t_n)$ or the binary-presence of the element t_n in $\tau(V_{a_i})$ (i.e., $v_{in}=1$ if $t_n \in \tau(V_{a_i})$, 0 otherwise). For example, say that the value transformation function τ is the toker scale on function, and that the function to weight the vector elements is $t^{1} \circ bir$ ary-presence. Then, the attributes are represented as a matrix: rows concept to the attributes; the columns correspond to the possible to sense appearing in the profile collections; and each element v_{in} is either 1 of the to en t_n appear in the attribute a_i) or 0 (otherwise).

- (iii) The similarity measure. For each possible pair of a bributes $(a_j, a_k) \in (A_{\mathcal{P}_1} \times A_{\mathcal{P}_2})$, their profiles \mathcal{T}_j and \mathcal{T}_k are compared according to a similarity measure (e.g. Dice, Jaccard, Cosine). Notice that the similarity measure must be compatible with the attribute model representation; for instance, the Jaccard similarity cannot be employed with the TF-IDF weighting.
- (iv) **The clustering algorithm**. The a¹-------kes as input the attribute names and the similarities of their profile. and performs the non-overlapping partitioning of the attribute nam (See Section 3.1.1 for more details). Its output is called *attributes parts* or *ng*.

2.3. Meta-blocking

The goal of meta-blocking[12] is to restructure a collection of blocks, generated by a redundant blocking technique, relying on the intuition that the more blocks two profiles share, the more likely they match.

Definition 2. META-BLC KING. Given a block collection \mathcal{B} , meta-blocking is the task of restructuring 'be set of blocks, producing a new block collection \mathcal{B}' with significar by higher precision and nearby identical recall, i.e.,: $\operatorname{precision}(\mathcal{B}') \gg \operatorname{precis}(\mathcal{M}') \cong \operatorname{recall}(\mathcal{B}') \simeq \operatorname{recall}(\mathcal{B}).$

In graph-based methods weighted graph $\mathcal{G}_{\mathcal{B}}\{V_{\mathcal{B}}, E_{\mathcal{B}}, \mathcal{W}_{\mathcal{B}}\}$ called **blocking** graph. V is the set of nodes representing all $p_i \in \mathcal{P}$. An edge between two entity profiles exists if they appear in at least one block together: $E = \{e_{ij} : \exists p_i, p_j \in \mathcal{P} \mid |\mathcal{B}_{ij}| > 0\}$ is one set of edges; $\mathcal{B}_{ij} = \mathcal{B}_i \cap \mathcal{B}_j$, where \mathcal{B}_i and \mathcal{B}_j are the set of blocks containing ρ_i and p_j respectively. $\mathcal{W}_{\mathcal{B}}$ is the set of weights associated to the edges. Methods weight the edges to capture the matching likelihood of the profiles that they connect. For instance, **block co-occurrence** frequency (e.k.a. CBS) [8, 18] assigns to the edge between two profiles p_u and p_v a veight ethor is to the number of blocks they shares, i.e.: $w_{uv}^{CBS} = |\mathcal{B}_u| \cap |\mathcal{B}_v|$. Then, edge-round strategies are applied to retain only more promising ones. Thus, at the end of the pruning, each pair of nodes connected by an edge forms a new block of the final, restructured blocking collection. Note that metable does in the restructured blocking collection. Note that metable does in the restructured blocking collection. Note that metable does in the restructured blocking collection. Note that metable does in the restructured blocking collection. Note that metable does in the restructured blocking collection. Note that metable does in the restructured blocking collection. Note that metable does in the restructured blocking collection. Note that metable does in the restructured blocking collection. Note that metable does in the restructured blocking collection. Note that metable does in the restructured blocking collection. Note that metable does in the restructured blocking collection. Note that metable does in the restructured blocking collection.



Two classes of pruning criteria can be employed ... met. 'nocking: cardinality**based**, which aims to retain the top-k edges, allowing a a-priori determination of the number of comparisons (the aggregate ca. 'inality and, therefore, of the execution time, at the expense of the recall; an ⁴ weight-based, which aims to retain the "most promising" edges through a weight threshold. The scope of both pruning criteria can be either *node-c*. *rtric* or *global*: in the first case, for each node p_i the top- k_i adjacent edges (or the edges below a local threshold θ_i) are retained; in the second case, the ι v-r \ldots es (or the edges below a global threshold Θ) are selected among the whole set of edges. The combination of those characteristics leads to four personal e_{μ} uning schemas: (i) Weight Edge Pruning (WEP) discards all the edges with weight lower than Θ ; (ii) Cardinality Edge Pruning (CEP) sort on in edges by their weights in descending order, and retains only the first K; (u_{i}) Weight Node Pruning (WNP [12]) considers in turn each node p_i and its adjacent edges, and prunes those edges that are lower than a local thre nold b_i (iv) Cardinality Node Pruning (CNP [12]) similarly to WNP is node on tric, 1 ut instead of a weight threshold it employs a cardinality threshold i_i (i.e., a_i ain the top- k_i edges for each node p_i).

3. The Blast Approac.

The main goals on *Plast* are: to provide an efficient, scalable and automatic method to extract loose schema information from the data; to perform a holistic combination of blocking and meta-blocking for ER exploiting this loose schema information.

These a e the n. in novelties w.r.t. other existing meta-blocking techniques, which are completely schema-agnostic [8, 12, 13].

Our ap_{F} ach akes as input two profile collections, and automatically generates \therefore block \bigcirc llection. It consists of three main phases, as schematized in Figur 4: loo. \therefore schema information extraction, loosely schema-aware blocking, and lo sely s nema-aware meta-blocking. In the following we give a high-level d'oription of each phase, then we dedicate a subsection for the details of each phase in urn.

Phase 1) The loose schema information is extracted. It consists of: the *atiributes partitioning*, and the *aggregate-entropy*. The former describes how the attributes are partitioned according to the similarity of their values; it is the result of the *attribute-match induction* task (Section 2...) The latter is a measure associated to each cluster of attributes, defined from the attribute entropies. We also introduce a *Locality-Sensible Hashing* (LSH) [19] optional step to reduce the computational complexity when dealing with data sources characterized by a high number of attributes.

- **Phase 2)** A traditional schema-agnostic blocking techni, ue is en lanced by exploiting the *attributes partitioning* to disambiguate key coording to the attribute partition from which they are extracted is particular, *Blast* employs Token Blocking, and we call the derived method *Loose Schema Blocking*.
- Phase 3) A graph-based meta-blocking is applied to the block collection generated in the previous phase. In particular, Γ'_{3st} meta-blocking exploits the aggregate entropy to weight the blocking group. The basic idea is the following. Each edge in the blocking grap. is associated to a set of blocking keys. Each blocking key is associated to an attribute. Each attribute has an information content that can be me sured through its entropy. Hence, the weight of an edge can be proportional to the information content of its associated attributes. For incharge, consider independent datasets containing records about people (as in Figure 1). Generally the attribute year of birth is less informative up in attribute name. This is because the number of distinct values of the *i*, rmer is typically lower than that of the latter. In fact, it is more "bely that two people are born in the same year, than they have the same name. Blast tries to assess the attribute information content employing the Shannon entropy, and assigns a weight to each blocking key proportional to the entropy of the attribute from which it is derived. T¹ as, ' sing Blast, records that share values of their name attributes are 'nore 'kel' indexed together than those sharing only values of their year f birth auributes. This process is completely unsupervised.

3.1. Loose Sche Information Extraction

(Phase 1 i ϵ Fie are 4) In Blast, the loose schema information extraction is performed throug an entropy extraction criterion applied in combination with the Loose uttribute-Match Induction, an attribute-match induction technique presented her. Moreover, we propose an optional LSH-based step for guaranteeing scale 'il' y on large datasets, which is the main improvement w.r.t. Attribute Cl stering [7].

3.1 1 L. attribute-Match Induction (LMI)

Follo ing the definitions of Section 2.2, Loose attribute-Match Induction (MI) is composed of these four components: the *tokenization* as value transformation function; the *binary-presence* of a token as weight for the attribute

representation model; the *Jaccard* coefficient as similarity measure and Algorithm 1 for clustering, a variation of the one introduced as *Attrib* te C_i , stering (AC) in [7].

Basically, Algorithm 1 first collects the similarities of all r ossible attribute profile pairs of two profile collections, and their maximum values of similarity (lines 2-8). The *similarity* function (line 4) measures the Jacca. ¹ coefficient:

$$jaccard(\mathcal{T}_i, \mathcal{T}_j) = \frac{\mathcal{T}_i \cdot \mathcal{T}_j}{|\mathcal{T}_i|^2 + |\mathcal{T}_j|^2 - \mathcal{T}_i \cdot \mathcal{T}_j}.$$

where $\mathcal{T}_i, \mathcal{T}_j$ are the vectors representing the at ribut a_i, a_j respectively (see Section 2.2).

Then, (lines 9-13) LMI marks as candidate match of the attribute each attribute that is "nearly similar" to its most similar at ribute by means of a threshold α (e.g.: $0.9 \cdot maxSimValue$). If an ottribute a_i has attribute a_j among its candidates, then the edge $\langle a_i, a_j \rangle$ is collected. Finally, the connected components of the graph built with these edges with cardinality greater than one, represent the clusters (line 14). Optice ally, a glue-cluster can gather all the singleton components (i.e., components that have cardinality equal to one), as in [7], to ensure the inclusion of all the pressure tokens (blocking keys).

Algorithm 1 Loose attribute-Match un ¹uction (LMI)

```
Input: Attr. names: A_{\mathcal{P}_1}, A_{\mathcal{P}_2}; Attr. profiles: \mathcal{I}_1, \ldots, \mathcal{T}_z; threshold: \alpha
Output: Set of attribute names clusters.
   1: edges \leftarrow \{\} sim \leftarrow Map\langle K, V \rangle
                                                                                  for each attr.
   2.
       Max \leftarrow Map\langle K, V \rangle // \text{ most similar attr.}
  3: for each a_i \in A_{\mathcal{P}_1}, a_j \in A_{\mathcal{P}_2} de

4: sim.push(\langle\langle a_i, a_j \rangle, simil \quad ity(\tau, \mathcal{T}_j) \rangle)

5: if sim.get(\langle a_i, a_j \rangle) > M \ x.get(a_i) then
   6:
                    Max.push(\langle a_i, sim \rangle)
   7:
              if sim.get(\langle a_i, a_j \rangle) > Max.g_{\neg} / \eta, then
                    Max.push(\langle a_j, s \ n \rangle)
   8:
  10:
 11:
 12:
              if sim.get(\langle \sigma , a_j \rangle) = (\alpha \cdot Max.get(a_j)) then
 13:
                   edges \leftarrow edges \cup \langle a_j, a_i \rangle
 14: K \leftarrow getCon^r sete ComponentsGrThan1(edges)
 15: return K
```

LSH-base ' Le se e tribute-Match Induction

The completed on of the similarity of all possible pairs of attribute profiles has an ov rall time complexity of $\mathcal{O}(N_1 \cdot N_2)$, where N_1 and N_2 are the cardinality of $A_{\mathcal{P}}$ and $A_{\mathcal{P}_2}$, respectively. For the dimensions commonly involved in the semi-structure ed data of the Web (the data sources schema can commonly have ϵ cen the sands of attributes) this is infeasible. However, only a few (or none) similar at ributes are expected to be found similar for each attribute; therefore, $em_{\mathcal{P}_2}$, ing techniques able to group the attributes approximately on the basis of the miliarity can significantly reduce the complexity of the attribute-match inductions, without affecting the quality of the results. Hence, $\frown B$ as we introduce a pre-processing step that can be optionally employed with be \frown LMI and AC.

LSH (*Locality-Sensitive Hashing*) allows to reduce the dimensionality of a high-dimensional space, preserving the similarity distances reducing significantly the number of the attribute profile comparisons. Employing the attribute representation model of LMI⁶ and Jaccard similarity, *Mir Hashing* and *banding* [20] can be adopted to avoid the quadratic complexity of comparing all possible attribute pairs.

The set of attributes is represented as a matrix, where each column is the vector \mathcal{T}_j of the attribute a_j (see section 2.2). Permuting the rows of that matrix, the minhash value of one column is the dement of that column that appears first in the permuted order. So, applying a set of n hashing function to permute the rows, each column is represented as a vector of n minhash; this vector is called minhash signature. The probability on definition to the same minhash value for two columns, permuting their rows, is equal to the Jaccard similarity of them; thus, MinHashing preserves the dimension of the vectors representing the matrix, with the advantage of reducing the dimension of the vectors representing the attributes. However, even for relatively dimension of the similarity of all possible minhash signature pairs may be dimensionally expensive; therefore, the signatures are divided into bands, and and signatures identical in at least one band are considered to be candidate pairs and given as input to the attribute-match induction algorithm (ada definition definition) to the set only through these candidate pairs - instead of all possible pairs).

Considering *n* minhash values as signature, *b* bands for the *banding* indexing, and r = n/b rows for band the p, bability of two attributes being identical in at least one band is $1 - (-s^r)^b$. This function has a characteristic *S*-curve form, and its inflection point \cdot r esents the threshold of the similarity. The threshold can be appr xim ted to $(1/b)^{1/r}$. For instance, choosing b = 30 and r = 5, the attribute part of that nave a Jaccard similarity greater than ~0.5 are considered for the *t* stribute match induction. (example Figure 5).

Thus, LSH call be employed as pre-processing step, before executing Algorithm 1, for filtering out attribute pairs that are most likely not similar, i.e., under a certain the shold⁷. Furthermore, minhash values can be employed for efficiently estimation of the Jaccard similarity [20] of two attributes (line 4 in Algorithm 1) Blast 1 flows this approach and stores minhash values in an array, which dor unations the space complexity of Algorithm 1. The space complexity of such an array is $\mathcal{C}(n \cdot (N_1 + N_2))$, where *n* is the number of minhash values, and N_1 and N_2 are the cardinalities of $A_{\mathcal{P}_1}$ and $A_{\mathcal{P}_2}$, respectively; thus, Algorithm 1 has a $\mathcal{O}(n \cdot (N_1 + N_2))$ space complexity.

⁶The L II attribute representation model can be used with Attribute Clustering [7] as well. ⁷For ou experiments we found that a threshold of 0.4 works well for all the dataset, but very ower thresholds can be employed; see Section 5.6 for experiments about the LSH "breshold.



Figure 5: The depicted curve represents the probabil. \cdot of two attributes to be considered "similar" (y-axis) in function of their actual similar 'ty (x-axis), when LSH is employed (with the parameters r=5 and b=30).

Entropy Extraction

To characterize each attribute cluster ς related during the attribute-match induction, *Blast* employs the Shannon *entropy* f its attributes. The entropy of an attribute is defined as follows [21]:

Definition 3. ENTROPY. Let X b on at vibute with an alphabet \mathfrak{X} and consider some probability distribution $p_{\chi^{(r)}} \circ f X$. We define the entropy H(X) by:

$$H(X) = \sum_{x \in \mathcal{I}} p(x) \log p(x)$$

Intuitively, entropy reprisents γ measure of *information content*: the higher the entropy of an attribute the mor significant is the observation of a particular value for that attribute. In other vords, if the attribute assumes *predictable* values (e.g., there are only 2 eo iprobable values), the observation of the same value in two different entity profiles coes not have a great relevance; on the contrary, if the attribute has more and redictable values (e.g., the possible equiprobable values are 100), of soming two entity profiles that have the same value for that attribute can be considered a more significant clue for entity resolution.

For example, co-sidering the data source 1 of Figure 1(a), the probability for a tuple to have as ε tribute Name the value "Ellen", i.e., p ("Ellen"), is 2/3 = 0.67, while the probability of having "John jr" as value is 1/3 = 0.33; thus, the entropy for the attribute Name is:

$$H(N me) = -p("Ellen") \cdot \log p("Ellen") - p("John jr") \cdot \log p("John jr") = 0.63$$

While, be stropy of the attribute Surname is 1.1, since all the tuples have c'iferent values for that attribute:

$$Tr(Surname) = -p("Abraham") \cdot \log p("Abraham") - p("Smith") \cdot \log p("Smith")$$

-p("Simons" $) \cdot \log p($ "Simons") = 1.1

in this case p(x) = 1/3 = 0.33.

In Blast the importance of a blocking key is proportional to be en. ppy of the attribute from which it is derived. This is obtained weighing in blocking graph according to the entropies (shown in section 3.3.1). To be a neutropy value for each group of attributes is derived by computing the aggregate entropy. The aggregate entropy of a group of attributes C_k is defined as:

$$\bar{H}(C_k) = \frac{1}{|C_k|} \cdot \sum_{A_j \in C_k} H(A_j) \tag{1}$$

When a schema-agnostic blocking (e.g. Token $\operatorname{Plocking}$) is applied in combination with attribute-match induction, each blocking key b_i is uniquely associated with a cluster C_k , $b_i \mapsto C_k$. For instance, considering the example of Figure 1(b), the token "Abram", disambiguated with distribute-match induction, can represent either the blocking key "Abram_c1" as pointed with the cluster C_1 , or the blocking key "Abram_c2" associated with the cluster C_2 ; where C_1 is composed of the attributes Name of p_1 and FullName of p_3 , while C_2 is composed of the attributes addr. of p_2 and Address of p_4 .

For meta-blocking, Blast employs $h(\mathcal{B}_j)$ the entropy associated with a set of blocking keys \mathcal{B}_j :

$$h(\mathcal{B}_j) = \frac{1}{|\mathcal{B}_i|} \sum_{b_i \in \mathcal{B}_j} h(b_i)$$
(2)

where $h(b_i) = \overline{H}(C_k)$ is the just γ associated to a blocking key $b_i \mapsto C_k$.

3.2. Loosely Schema-aware L'acki g

(Phase 2 in Figure 4) If Blast we employ Token Blocking, as in [7]. Other blocking techniques (e.g., employing q-grams instead of tokens, as in [22]) can be adapted to this slope as will, but comparing them is out of the scope of this paper. For sake of plotsentation, we call **Loose Schema Blocking** the combination of Loose attribute Match Induction and Token Blocking. The results is that each toker (i.e., blocking key) can now be disambiguated according to the cluster of the ottribute in which it appears, while in classical Token Blocking each token represents a unique blocking key. The example in Figure 2 gives an intuition cluber catribute in which it appears avoids to index together some non-method in a present should be blocking graph weighting, and, at the end cluber the method is a void one superfluous comparison.

3. [°] Loosing Schema-aware Meta-blocking

(Phase 3 in Figure 4) Blast introduces a novel node-centric meta-blocking technique designed to exploit *loose schema information*.

Papadakis et al. [12] demonstrated that node-centric blocking-graph pruning criteria (i.e., WNP and CNP) outperforms the global ones (i.e., WEP and CEP),

	p_v (p_3)	$\neg p_v (\neg p_3)$	
$p_u(p_1)$	n_{11} (4)	$n_{12}(2)$	n_{1+} (6)
$\neg p_u (\neg p_3)$	n_{21} (3)	n_{22} (3)	n_{2+} (6)
	n_{+1} (7)	n_{+2} (5)	$n_{++} (12)$

Table 1: Contingency table for p_u , p_v . In parentheses an example 'original from blocks in Figure 1(b).

and that *weight-based* pruning criteria outperform the *care malit j-based* ones in terms of recall, but at the expense of precision. Loo e schema information can be exploited to significantly enhance precision; for t. is easo , and considering the aforementioned results achieved by [12], as a d night choice, Blast employs a weight-based, node-centric pruning criterion (i.e., WN.).

In the following the two steps of Blast met. blocking are described. In the first step, the blocking graph $\mathcal{G}_{\mathcal{B}}\{V_{\mathcal{B}}, E_{\mathcal{B}}, \mathcal{W}_{\mathcal{B}}\}$ is generated weighting the edges according to a weighting schema designed to capture the relevance of the profiles co-occurrence in the blocks, and to exploid the attribute entropies. The second step consists in a novel pruning criterion.

3.3.1. Blocking Graph Weighting

Considering two entity profiles $p_u \, c \, d \, p_v$ the contingency table, describing their joint frequency distribution in a green clock collection, is shown in Table 1. The table describes how entity prover and p_v co-occur in a block collection. For instance: the cell n_{12} represents the number of blocks in which p_u appears without p_v (the absence is denoted with "¬"); the cell n_{2+} represents the number of blocks in which p_u is not present (independently of p_v). These values are also called *observed* values. A an example, the values in parentheses are values derived from the block collection if Figure 1(b) for the profiles p_1 and p_3 .

Given this representation, Blast employs Pearson's chi-squared test (χ^2) [23] to quantify the introduce of p_u and p_v in blocks; i.e., testing if the distribution of p_v , since that p_u is present in the blocks (first row of the table), is the same as the distribution of p_v , given that p_u is not present (the second row in the table). In practice, the chi-squared test measures the divergence of observed (n_{ij}) and expected (μ_{ij}) sample counts (for i = 1, 2, j = i, 2). The expected value are with reference to the null hypothesis, i.e., assuming that p_u and p_v appear independently in the blocks. Thus, the expected value for each cell of the contangency table is: $\mu_{ij} = (n_{i+} \cdot n_{+j})/n_{++}$.

Hence, \therefore wei, ht w_{uv} associated to the edge between the nodes representing the entry product p_u and p_v is computed as follows:

$$w_{uv} = \chi_{uv}^2 \cdot h(\mathcal{B}_{uv}) = \sum_{i \in \{1,2\}} \sum_{j \in \{1,2\}} \frac{(n_{ij} - \mu_{ij})^2}{\mu_{ij}} \cdot h(\mathcal{B}_{uv})$$
(3)

Notice that Blast uses the test statistic as a measure that helps \circ highlight particular profile pairs (p_u, p_v) that are highly associated in the block collection, and not to accept or refuse a null hypothesis. The correcting entropy relue just weight the importance of the blocks in which a co-occurrence oppear, since not all the blocks are equally important (as discussed in section (1.1)).

3.3.2. Graph Pruning

Selecting the pruning threshold is a critical task. We lentify a fundamental characteristic that a threshold selection method, in WNF, \cdots , present: the independence of the local number of adjacent edges, to foid the sensitivity to the number of *low-weighted* edges in the blocking \circ aph. In fact, this issue arises when employing threshold selection function, that the pend on the number of edges, such as the *average* of the weights [12]. To illue trate this phenomenon,



Figure 6: (a) Two additional profiles $\gamma t_{\rm theo}$ llection in Figure 1; (b) the node-centric representation of the blocking graph for $_{\rm F}$

consider again the example in Figure 6. Figure 6(b) shows \mathcal{G}_{p_1} , the node-centric view of the $\mathcal{G}_{\mathcal{B}}$ for the property p_1 .

If the profile collection (as . Figure 1(a)) is composed only of the profile set $\{p_1, p_2, p_3, p_4\}$, the resulting graph \mathcal{G}_{p_1} has only 4 nodes and 3 edges. In this scenario the average of t'e edge weights (the local pruning-threshold) is slightly greater than 2. Thus, only the edge between p_1 and p_3 is retained in the pruning phase. Let if the two entity profiles in Figure 6(a) are added to the profile collection, then two nodes and two edges are added to \mathcal{G}_{p_1} . This influences the thre hold that became 1.8. Consequently, the edge between p_1 and p_4 is retained in the pruning phase. Therefore, the comparison of p_1 and p_4 depends or the presence or absence of p_5 and p_6 in the profile collection, even though the similarity between those two profiles does not depend on p_5 and p_6 .

In Blast ve ir roduce a weight threshold selection schema independent of the number of edges in the blocking graph.

Loca Thres old Selection. In the node-centric view of the blocking graph, the edg, wit¹ the highest weight represents the upper bound of similarity for the combination of the underlying blocking technique and weighting function; so, ve proper to select a threshold independent of the number of adjacent edges



Figure 7: Weight threshold. A directed edge from $p_i \circ p_j$ indicates that the weight of the edge e_{ij} is higher than θ_i ; a directed edge from $p_i \circ p_j$ indicates that the weight of the edge e_{ij} is higher than θ_j .

by considering a fraction of this upper by ur 1:

$$\theta_i - \frac{M}{c}$$
 (4)

where M is the local maximum right, and c an arbitrary constant. A value for c that has shown to be efficacious with real dataset is c=2; a higher value for c can achieve higher recell, but at the expense of precision.

Having determined the local thr shold for each node, the last step to perform is the retention of the edges. The light in node centric pruning, each edge e_{ij} between two nodes p_i and γ_j is related to two thresholds: θ_i and θ_j (Figure 7(a)); where θ_i and θ_j are the breshold associated to p_i and p_j , respectively. Hence, as depicted in Figure (b), each edge e_{ij} has a weight that can be: (i) lower than both θ_j and θ_j , (ii) higher than both θ_i and θ_j , (iii) lower than θ_i and higher than v_j , or (1) higher than θ_i and lower than θ_j . Cases (i) and (ii) are not ambiguous therefore e_{ij} is discarded in the first case, and retained in the second or . But, cases (iii) and (iv) are ambiguous.

Existing met. clocking papers [12] propose two different approaches to solve this ambig ity: $rede_jined$ WNP retains e_{ij} if its weight is higher than at least one of the two the esholds (i.e., a logical disjunction, so we cal this method WNP_{OP}). while *eciprocal* WNP retains the edge if it is greater than both the threshold (i.e., a logical conjunction, so we cal this method WNP_{AND}). Here in Blast we choole to employ a unique general threshold, equals to:

$$\theta_{ij} = \frac{\sqrt{\theta_i + \theta_j}}{d} \tag{5}$$

where d is a constant; for d = 2 the resulting threshold θ_{ij} is equal to the matrix of the two involved local threshold, and has shown to perform well with

real datasets.

The experimental section 5.3 shows how the parameters c ard d , ^quence the performances of *Blast* and in particular, the tradeoff of precision and recall for an ER task.

4. Distributed Meta-blocking

We now introduce basic concepts of MapReduce-like systems and then describe what is needed to parallelize *Blast* for taking full adva. and of parallel and distributed computation.

4.1. MapReduce-like Systems

In MapReduce-like Systems, programs are written in functional style and automatically executed in parallel on a cluster of pachicles. These systems also provide automatic mechanisms for load balanch grand to recover from machine failures without recomputing the whole program by veraging on the functional programming abstraction (e.g., *lazy evaluation* in Apache Spark [24]). In the following, we present the main *function* employed to formalize MapReduce-like algorithms in this paper with a concise and *spark-like* syntax. These functions are defined w.r.t. *Resilient Distributed Datestet* (RDD [24]), which are the basic data structure in Apache Spark. In packet is a distributed and resilient collection of objects (e.g.: *integers, strings*, etc.).

Basic Functions for MapReduce-like A. prithms

- map (map in MapReduce [25]) applies a given function to all elements of the RDD returning a new BDD.
- mapPartitions: appl'es a g- "er function to each RDD partition returning a new RDD.
- reduceByKey (reduce $\ M$ pReduce [25]) reduces the elements for each key of an RDD using a specified commutative and associative binary function.
- groupByKey: groups the values for each key in the RDD into a single collection.
- join: perfor. ~ a hash join between two RDDs.
- broadc: st: 'roadcasts a read-only variable to each node in the cluster (which cache iv)

We enploy this set of functions for the sake of presentation of our algorithms for MapReduce-like systems (Section 4.2). Yet, the algorithms discussed in this $p^{e_{pred}}$ employing such functions are general enough to run on any MapReduce-like systems.

In M^r pReduce-like systems implementations, functions like join and groupByKey are notoriously expensive, due to the so-called shuffling of data across the $n\epsilon$ -work [26]. In fact, they involve redistribution of the data across partitions

with the consequent overheads: data serialization/deserialization, toost ission of data across the network, disk I/O operations. For instance, ioin implies that all the records that have the same key are sent to the same node. Whereas, map and mapPartitions are usually fast to compute, because data is locally processed in memory, and no shuffling across the network is equived [26].

4.2. Blast on MapReduce-like Systems

4.2.1. Distributed Blocks Generation

For the loose information extraction and loosely schematic blocking (Phases 1 and 2 in Figure 4), adapting the proposed solution of Seman 3 to the MapReduce paradigm is straightforward. It only requires an order ying MapReduce-based LSH algorithm (such as [27]). Then, adapting from Blocking to the MapReduce paradigm is straightforward as well (it essent ally builds an inverted index).

The main challenge for the parallelization of *Blast* is related to the graphbased meta-blocking step. In fact, the blocking graph, defined in Section 2, is an abstract model useful to formalize and devise meta-blocking methods. However, materializing and processing the whole blocking-graph may be challenging in the context of big data due to the size of such a graph. For this reason, algorithms for processing the blocking-graph have bee, proposed to scale meta-blocking to large datasets on MapReduce-like system [1.]. Their basic idea is to distribute the blocking-graph processing on multiple machines, trading a fast execution for high resource occupation.

In the following, firstly we revise the state-of-the-art blocking-graph processing algorithm, i.e., $repartit^{i}$ meta-blocking⁸[13], discussing its limitations; then, we present our novel algorithm called *broadcast meta-blocking*, which overcome these limitations.

4.2.2. Distributed Blc kine grav's Processing

Repartition met: block. ,— At the core of repartition meta-blocking [13] there is a full mat_radiation of the blocking graph.

Algorithm 2 describes the repartition meta-blocking with pseudocode. Firstly, for each profil and for each of its blocking key, a pair $\langle key, profile \rangle$ is generated (Lines 3 1). The result can be seen as a table P^K with two columns: key and profile. Then, a self-join on P^K (Line 6) and a group by profile (Lines 7) are performed. In practice, this corresponds to a graph materialization, since each node the second with a copy of its local neighborhood. As a matter of fact, end, where p_i is fixed and p_j is a profile sharing at least one blocking key with p_i .

⁸In [1.] this algorithm is called *entity-based parallel meta-blocking* (an example is shown i. Figure 4 of [13]) and it is the state-of-the-art (i.e., fastest and efficient) algorithm for performing node-centric pruning on the blocking graph; we coined the term repartition meta-blocking for the analogy with the repartition join algorithm [28].



Finally, for each profile p_i and its neighborhood (Lines 8-10), a pruning function computes a local threshold θ_i and the only the edges with a weight higher than θ_i (Lines 9)⁹.

Optimization note— When implementing rescalition meta-blocking, for alleviating the network communication bottlene, ', blocks and profiles are represented by their ids, as proposed in [13]. This means that, for Algorithm 2, the pair $\langle key, profile \rangle$ (in Line 5) is a pair of identifiers: the first id represents the key (i.e., the block), the second id represent he entity profile.

Example 1. An example of the execution steps of repartition meta-blocking is shown in Figure 8. Five profiles we grouped in three partitions: $\{p_1\}, \{p_2; p_3\}$ and $\{p_4; p_5\}$. Each partit on is as igned to a worker (i.e., a physical computational node) that computer is a signed to a worker (i.e., a physical computational node) that computer is a signed to a worker (i.e., a physical computational node) that computer is a signed to a worker (i.e., a physical computational node) that computer is a signed to a self join in order to yield the bag of all the comparison part p_i, p_i ; this step (Step 2) requires a shuffling of the data (P^K) through the new red (note that only the ids of the profiles are sent around the networ is). The comparison pairs are assigned to a worker according to their keys, so the group by operator partitions them to materialize the neighborhoods within each worker (Step 3). Thus, in parallel, each neighborhood can be processed is are erate the final restructured block collection (Step 4).

The bostler sck of repartition meta-blocking is the join (Line 6 in Algorithm 2). In fact, F thyr iou et al. [13] describe it as a standard repartition join [28] (a.k.a. $\neg duce$, i.e. join), a notoriously expensive operator for MapReduce-like systems¹⁰. A workaround for this issue could be the employment of broadcast

⁹Some pruning functions requires as input both the local threshold of the current node p_i and the local threshold of its neighbors; in this case, (Lines 8-10) are executed two times: first, for prompting all the thresholds (which are then broadcasted); then, for the actual pruning.

¹⁰ We make explicit the *join* operator: Efflymiou et al. present their algorithms in [13] by us \cdot_{5} only map and reduce functions.



Figure 8: Repartition 1. eta- 1. king example

join [28], a join operator for MapRe use like systems that is very efficient if one of the join tables can fit in main memory. Unfortunately, P^{K} (Line 6 in Algorithm 2) typically cannot is in memory with large dataset (e.g., those employed in our experiments in Section 5). Thus, broadcast join cannot be employed in Algorithm 2.

Broadcast meta-blocki. To void the repartition join bottleneck, we propose a novel algorithm for para. "I neta-blocking inspired by the broadcast join. The key idea of our algorithm is the following: instead of materializing the whole blocking graph, only a portion of it is materialized in parallel. This is possible by partitioning the not. of the graph and sending in broadcast (i.e., to each partition) all the information needed to materialize the neighborhood of each node one at a time. Once the neighborhood of a node is materialized, the pruning functions that can be applied are the same employed in *repartition meta-blocking* [13], and (non-parallel) *meta-blocking* [12, 8].

The pse docod of broadcast meta-blocking is shown in Algorithm 3 and described in the following. Given the profile collection \mathcal{P} the block index I_B is generated Lines 1-2): it is an inverted index listing the profile ids of each block (Clocks a. represented through ids as well). When executing Blast, the functions bun 'Blocks and buildBlockIndex also extract the loose schema informatio. —i.e., they basically perform what is described in Section 4.2.1. Then, I_T is broadcasted to all workers (Line 4), in order to make it available to them. (In each partition, an index I_P is built (Lines 5-6): for each profile it lists the block identifiers in which it appears. Then, for each partition and for each profile, by using the I_P and I_B indexes, a profile's neighborhood at a time is built lo any (Lines 7-9): for each block id contained in I_P it is possible to obtain

Algorithm 3 Broadcast Meta-blocking

Input: *P*, the profile collection **Output:** C, the list of retained comparisons 1: $B \leftarrow buildBlocks(P)$ 2: $I_B \leftarrow buildBlockIndex(B)$ 3: $C \leftarrow \{\}$ // retained comparisons 4: **broadcast** (I_B) 5: map partition $\langle part \rangle \in P$ $I_P \leftarrow buildProfileBlockIndex(I_B)$ 6: 7:for each $profile \in part$ do $B_{ids} \leftarrow I_P[profile.id]$ 8: $profileNeighborhood \leftarrow buildLocalGraph(B_{ids, -B})$ 9: $C_p \leftarrow \text{prune}(profileNeighborhood})$ 10: $C.append(C_p)$ 11:



Figu. 9: Broadcast meta-blocking example

from I_B the l₁ t₀ of profile ids (the neighbors). Finally, it performs the pruning (Lines 10-1¹)¹¹.

Note t'.at the prune function employed in Algorithm 2 (Line 9) and Algorithm . (Line 1) takes as input a profile's neighborhood and can be any node-centric p. Ing function, e.g., the one described in Section 3.3.

Example 2. An example of the execution steps of broadcast meta-blocking is shown. Finare 9. In Step 1 the profiles are partitioned and assigned to the i orkers. Then, in Step 2, the inverted index of blocks (the Block Index) is

 $^{^{11}}$ As for Algorithm 2, for some pruning functions, this last iteration has to be performed tw ce: the first time for computing all the thresholds, the second for the actual pruning.

built—for the sake of the example, the intermediate steps to build be inverted index are not depicted. This step requires a shuffling of data though be network, but at a significantly lower extent compared to that needed for unself-join operation of repartition meta-blocking. Then, the Block Index is upper orderasted to all the workers that perform the last phase of the processing Ster 2). Finally, in Step 3, each worker processes a partition of the profile set: upper anterializes a neighborhood at a time by exploiting the local instance of the Block Index, and performs pruning to yield the final restructured block coluction.

5. Evaluation

The experimental evaluation aims to answer the following questions:

- Q1: What is the performance of Blast in terms of precion, recall, and execution time compared to the state-of-the-art [12]: (section 5.1)
- Q2: What is the contribution of each Bl. component to the overall performance (e.g., how the performance change. by employing the aggregate entropy)? (Section 5.2)
- Q3: What are good parameters c a: ' d for 'he pruning threshold of Blast (see Section 3.3.2) for a good recall/p.ec. 'on tradeoff? (Section 5.3)
- Q4: How efficient is broadcast n. *a-Diocking, compared to repartition metablocking [13]? (Section 5.4)
- Q5: How does Blast (with broad ast meta-blocking) scale when varying the number of machines a vailable for the ER processing? (Section 5.5)
- Q6: How does the LSF -bas d step affects the Blast processing? (Section 5.6)
- Q7: What is the performer of Blast w.r.t. traditional meta-blocking when no schema-aligner at is required (i.e., with a single data source with known schema containing 'multicates)? (Section 5.7)
- Q8: What is he erformance of Blast w.r.t. traditional meta-blocking in a multi-data arce context (i.e., when the number of data sources is greater than ?)? (Secu. n 5.8)

Experimenta. Set .p

Hard ware : nd Software—All the experiments are performed on a ten-node cluste : each 1 ode has two Intel Xeon E5-2670v2 2.50 GHz (20 cores per node) and 126 CP of RAM, running Ubuntu 14.04. All the software is implemented i . Scala ?.11.8 and available at [29]. To assess the performance of the state-oft ve-art n sta-blocking methods we re-implemented all of them for running on Apa.¹ Spark as well. We employ Apache Spark 2.1.0, running 3 executors on

	Size	$ \mathcal{P}_1 - \mathcal{P}_2 $	$ \mathcal{A}_1 - \mathcal{A}_2 $	$ \mathcal{D}_P $
articles1(*)	small	2.6k - 2.3k	4 - 4	2.2k
articles2(*)	small	2.5k - 61k	4 - 4	2.3k
products $(*)$	small	1.1k - 1.1k	4 - 4	1.1 ^L
movies	small	28k - 23k	4 - 7	∕ 3k
articles3(*)	large	1.8M - 2.5M	7 - 7).6M
dbpedia	large	1.2M - 2.2M	30k - 50k	0
freebase	large	4.2M - 3.7M	37k - 11k	$1.5\mathrm{M}$

Table 2: Dataset characteristics: number of entity profiles, number of a tribute names, and number of existing matches. An exact schema alignment c = n be a nieved only on starred "(*)" datasets.

each node, reserving 30 GB of memory for the mask 'nous. We set the default parallelism to twice the number of cores as suggested by best practice¹².

Datasets—Table 2 lists the 7 real-world datase. employed in our experiments. They have different characteristics and are from variety of domains. The small datasets (i.e., articles1, articles2, products, and movies) are used only when evaluating the performance in terms of recall and precision, since their time performance on distributed setting 's p it significant. (Table 4 reports the definition of precision and recall from Sect. in 2.)

All the datasets match two different is a sources for which the ground truth of the real matches is known. From [30, articles1 matches scientific articles extracted from dblp.org and dl.ac. org, articles2 matches scientific articles extracted from dblp.org and scholar.g. gle.com. products matches products extracted from Abt.com are Prov.com. From [7]: movies matches movies extracted from imdb.com a d dbpe 'ia.org; dbpedia matches entity profiles from two different snapshots of Prodei (2007 and 2009)¹³. From [31]: articles3 matches scientific articles extracted from Citeseer and DBLP. Finally, freebase is derived from the B' for Frip'e Challenge 2012 Dataset [32]: it is composed by two datasets, one conterns the data of DBpedia 3.7, the other one the data of Freebase; we cleaned these two datasets keeping only the information in English, removing other 1: aguages; the ground truth is represented by the *owl:sameAs* relationships be end them.

Methods Con." arations and Results Analysis—For each dataset, the initial block collection ... extracted through a redundant blocking technique (either Token Blocking or Loose Schema Blocking). Then, the block collection is processed with Lock Purging and Block Filtering [12], which aim to remove/shrink the lagest blocks in the collection. Block Purging discards all the blocks that contain more than half of the entity profiles in the collection, corresponding to highly frequent blocking keys (e.g. stop-words). Block Filtering removes

²https://spark.apache.org/docs/latest/tuning.html

 $^{^{13}}$ Only 25% of the name-value pairs are shared among the two snapshots, due to the constant change in DBpedia, therefore the ER is not trivial.

each profile p_i from the largest 20% blocks in which it appears¹⁴ Tⁱ e time required by both Block Purging and Block Filtering is negligible compred to the meta-blocking phase, thus not listed in the experimental results.

The schema-agnostic meta-blocking methods can be executed on blocks generated with both Token Blocking and with Loose Schema Blocking, while *Blast* is compatible with the latter only, since it exploits the loose schement information.

For the schema-agnostic meta-blocking methods, we r port the average values of recall, precision, F1-score¹⁵ and time obtained by executing each method in combination with each of the five weighting schemas p_1 posed in [7]¹⁶. We also report that no traditional weighting schema ar 1 pruning strategy combination performs better than the other on the consilion discuss, confirming the results of [7].

Finally, for the time measurement, we report the values obtained by averaging the times recorded for five runs. Table 3 submarized the acronyms used in this Section.

5.1. Blast vs. State-of-the-art Meta-blocking

Table 3 summarizes the acronyms and configurations employed in this experiment. WNP and CNP is applied on block collections generated both with Token Blocking (TB) and Loose Schema L'ocking (LSB), and employing both redefined (WNP_{OR}/CNP_{OR}) and receiver at (WNP_{AND}/CNP_{AND}) approaches (see Section 3.3.2).

Figure 11 shows the result on the exact ution of Blast and traditional metablocking on all the datasets. Compared to WNP approaches, Blast achieves significantly higher precision and basically the same level of recall on all the datasets. In particular Bli st always outperforms LSB+WNP_{OR/AND}, demonstrating that the Blast weight +-base , pruning is actually more effective than the traditional ones.

Compared to TB+ $CNF_{OR/,ND}$, Blast achieves higher precision on all the datasets, with the evception constitutes and freebase, where CNP_{AND} has a higher precision Figure 11(i) and Figure 11(n)). Notice though that on articles2 and confrectes Blast achieves a recall significantly higher (Figure 11(b) and Figure 11(g)). On all the other datasets, the recall of Blast is almost the same of TB+CNP_{OR/AND} (Figure 11(a-g)), or slightly higher (Figure 11(b) and Figure 11(g)). Similarly, Blast outperforms LSB+CNP_{OR/AND}

 $^{^{14}}$ This heu. ic he shown to not affect recall in practice, while lighting the blocking-graph handlin [-2].

¹⁵ H and et a. [33] have recently discussed how F1-score may be an unreliable measure for compa ing differ nt ER algorithms. We report F1-score for the sake of completeness—it has been use ` in m ay related works [5, 34, 35]—yet we draw conclusions on the basis of precision ar a recall only.

¹⁶Amon, the weighting schemas proposed in [7], we did not identify an overall best performer a d an overall worst performer, confirming the results reported in [13], for this reason we report the separate precision, recall, F1-score and execution time.

ACCEPTED MANUSCRIPT

	Blocking
ТВ	Token Blocking [7] (see Section 1)
LSB	Loose-Schema Blocking (see Section 3.2)
	Meta-blocking
WNP	Weight Node Pruning [12] (see Section 2.3)
CNP	Cardinality Node Pruning [12] (see Section 2.3)
$WNP_{OR}(CNP_{OR})$	The <i>redefined</i> WNP (CNP) approach [1 ^{\circ}] (see Section 3.3.1). An edge is not pruned if it weight is grater t ¹ and y of its adjacent node's local thresholds (OK aution)
$WNP_{AND}(CNP_{AND})$	The reciprocal WNP (CNP) approa. [12] (see \mathfrak{S} tion 3.3.1). An edge is not pruned if it weight is gree. "than both of its adjacent node's local thresholds (AND co dition)
	Blast

$Blast_{\chi^2}$	Blast approach, without employing ' \circ aggregate entropy to compute the weights of 41 '.' Section 3.3.1).
$Blast^{\mathcal{H}}$	Blast approach, using the working schema proposed in [12] instead of χ^2 to weight the edges free Section 3.3.1). The entropy is used. The results reported for a schema considering schema.
$Blast_{\chi^2}^{\mathcal{H}}$ (or simply $Blast$)	Blast approach (i.e., with, and aggregate entropy, see Section 3).

Table 3: Acronyms and configurations.

$\ \mathcal{B}\ $	Number of comparisons ε_{\cdot} , ailed by a block collection ${\cal B}$
$ \mathcal{D}^{\mathcal{P}} $	Number of aup. Stes (matches) in a profile collection \mathcal{P}
$ \mathcal{D}^{\mathcal{B}} $	Number of duplicat ; (matches) indexed in at least one block $b \in \mathcal{B}$
$recall(\mathcal{B})$	$ \mathcal{D}^{\mathcal{B}_{+}} \mathcal{D}^{\mathcal{P}} $
$recision(\mathcal{B})$	$ \mathcal{T} / B $

Table 4: Metrics.

in terms of procision on all the datasets but articles2 and freebase (Figure 11(i) and 1 models in the latent datasets Blast yields a higher recall (Figure 11(s)) and 1 gure 11(g)).

We also considered the overall execution time of the methods. For the comparison, we could get our Spark implementation of them, employing broadcastmeta-'nocking as core blocking-graph processing algorithm, running on a single n de (for scalability and performance on multiple nodes see Section 5.5). In such a configuration, for the small datasets the results are not reported: $t^1 = overhead$ introduced by Spark in each execution does not allow to propcoly recond the actual time efficiency of such configuration when the size of



Figure 10: Execution time of the different methe 's ap_{P} ' , on blocks obtained with the Token Blocking (TB+WNP_{ADN/OR}/CNP_{ADN/OR}) and with the Loose Schema Blocking (LSB+WNP_{ADN/OR}/CNP_{ADN/OR}). The execution time is referred to the meta-blocking, and it was taken on a single node \sim the biggest datasets.

the data is small¹⁷. The results are Jown in Figure 10. Blast is always significantly faster than $CNP_{OR/AND}$ on all the considered datasets and all the configurations (up to $3.8 \times$ on diagonal figure 10(b)). It is also faster than TB+WNP_{OR/AND} on dbpedia ($2.8 \times$ in Figure 10(b)) and freebase ($1.6 \times$ in Figure 10(c)); while, on articles3 is slightly slower (Figure 10(a)). Compared to LSB+WNP_{OR/AND}, Bla a almost the same execution time on dbpedia (Figure 10(b)) and freetise (Figure 10(c)); while on articles3 is slightly slower (Figure 10(b)) and freetise (Figure 10(c)); while on articles3 is slightly slower (Figure 10(b)).

Overall, we conclude that Blast yields the same recall and a significantly higher precision of the best proforming schema-agnostic meta-blocking methods [12], on each dreaset — The only exception is LSB+CNP_{OR/AND}, which achieves higher relation that all blast on two of the seven considered datasets (Figure 11(i) and Figure 11(n_c)), but at the same time has lower recall (Figure 11(b) and Figure 11(c_c)) and is always slower than Blast Figure 10. Finally, we also observe that the transition of the fastest schema-agnostic method.

 $^{^{17}}$ In [11] the $_{\circ}$ ref or these datasets are reported for the Java implementation and the results are and ogous.

 $^{^{18}}$ T he differences between Blast and WNP/CNP are statistically significant according to Studen 's T-Tes (with p-value <0.05).



Four 11 Recall and precision achieved by the considered methods on all the datasets. Transional meta-blocking (WNP_{ADN/OR} and CNP_{ADN/OR}) has been combined both with "oken Blocking (TB+WNP_{ADN/OR}/CNP_{ADN/OR}) and Loose Schema Blocking ("SB+WNP_{ADN/OR}/CNP_{ADN/OR}). Blast is based on Loose Schema Blocking for the elementation of the loose schema information, thus it is not applicable on block collection generate with Token Blocking. 29

ACCEPTED MANUSCRIPT





In this experiment we evaluate 'ne contribution provided by each component characterizing Blast: the aggregate e. tropy and the weighting function. The results are reported in Figure 12.

We compare three different configurations of meta-blocking performed on a block collection generated through Loose Schema Blocking: $Blast_{\chi^2}$, $Blast_{\chi^2}^{\mathcal{H}}$, $Blast_{\chi^2}^{\mathcal{H}}$, as described in Table 5.

Aggregate Entropy

The comparisor of $Blast_{\chi^2}$ and $Blast_{\chi^2}^{\mathcal{H}}$ allows us to assess the contribution of the *aggregate entropy*. The result in Figure 12(h-n) shows that by employing the *aggregate entropy* precision increases from 1.6 (Figure 12(h)) to 3.7 times (Figure 12(n)'. A the same time, recall is almost the same on all datasets (Figure 12(a-g)). On **freebase**, $Blast_{\chi^2}^{\mathcal{H}}$ even achieves both recall and precision significant' *f* hicher than $Blast_{\chi^2}$ (Figure 12(g) and Figure 12(n)).

We conclude that aggregate entropy actually enhances meta-blocking.

Chi-s uared reighting

By stemp by a weighting function derived from the chi-squared (χ^2) statistical contractions are signed to quantify the significance of the co-occurrences (see Secton 3.3). For assessing the performance of this weighting function, $Blast^{\mathcal{H}}$ is compared with $Blast^{\mathcal{H}}_{\chi^2}$. The result is shown in Figure 12. Recall is almost one same for all the datasets for $Blast^{\mathcal{H}}$ and $Blast^{\mathcal{H}}_{\chi^2}$ (Figure 12(a-g)), while $Blast^{\mathcal{H}}_{\chi^2}$ achieves a considerably higher precision (Figure 12(h-n)), e.g. on

dbpedia (Figure 12(m)) precision has a $16 \times \text{improvement}$. The only exc ptions are articles2 and freebase: on the former, $Blast_{\chi^2}^{\mathcal{H}}$ achieves alreast upper same recall and precision yielded by $Blast^{\mathcal{H}}$ (Figure 12(b) and Figure 12(r)) on the latter, $Blast^{\mathcal{H}}$ has a 4.6% higher recall, yet $Blast_{\chi^2}^{\mathcal{H}}$ yields a provision more than twice higher than $Blast^{\mathcal{H}}$ (Figure 12(n)).

We conclude that our weighting function actually enhances is sta-blocking performance.

5.3. Blast sensitivity to parameters

From Section 3.3.2, to perform the graph pruni g, Γ_{iasi} computes a local threshold θ_i for every profile p_i . This local threshold is conjusted as $\theta_i = \frac{M}{c}$ (from Equation 4), where M is the local maximum wight, and c is an arbitrary constant. Then, for retaining an edge between two p offles p_i, p_j , a unique threshold θ_{ij} is computed as $\theta_{ij} = \frac{\sqrt{(\theta_i^2 + \theta_j^2)}}{d}$ (from Equation 5), where d is an arbitrary constant.

The constants c and d can be reduced to a unique constant $t = c \cdot d$, as shown below:



Figure 13: Blast sensitivity: these charts shown the variations of precision, recall, and F1 score in r_{c} ction of the t parameter.

$$\theta_{ij} = \frac{1}{c} \cdot \sqrt{\left(\frac{\theta_i}{d}\right)^2 + \left(\frac{\theta_j}{d}\right)^2} = \frac{1}{c} \cdot \sqrt{\frac{\theta_i^2}{d^2} + \frac{\theta_j^2}{d^2}} = \frac{1}{c} \cdot \sqrt{\frac{1}{d^2} \cdot \left(\theta_i^2 + \theta_j^2\right)}$$
$$= \frac{1}{c \cdot d} \cdot \sqrt{\theta_i^2 + \theta_j^2} = \frac{1}{t} \cdot \sqrt{\theta_i^2 + \theta_j^2}$$
(6)

We perform a preliminary experiment by varying t in the range (2–10) in order to choose the best values for c and d. Notice that it is not possible to set $t \leq 1$, otherwise $\theta_{ij} > max(\theta_i, \theta_j)$, so every edge will be pruned. For bermore, we limit $t \geq 2$ because, in practice, lower values of t yields very p for recall for many of the analyzed datasets.

The results are shown in Figure 13. In general, we observe bat the recall increases as t increases, but at the expense of precision. As a trade-off for precision and recall, for all the experiments in this paper, we employ t = 4 (setting c = 2 and d = 2). As a matter of fact, on all the a^{+asc} is, increasing t above 4 the loss of precision is traded for a little gait in t^{1} recall.

5.4. Broadcast vs. Repartition Meta-blocking

The goal of this experiment is to compare the efficiency of broadcast metablocking (Algorithm 2) and repartition meta-blocking (Algorithm 3). Both the algorithms can be employed as core graph-prodessing algorithms for any metablocking method. Thus, we evaluate them in combution with WNP and CNP, in order to analyze how they perform on the family of meta-blocking, i.e., those based on weight-threshold, and those based on cardinality-threshold (see Section 2.3). To minimize additional determan, we run them in combination with the computationally cheapest weight, g function, i.e., block co-occurrence frequency (we record analogues tren 's with other weighting functions). The experiment was performed on 10 node. We consider only the large datasets since the overhead introduced by "Data Loes not pay off on the small ones on multiple nodes. Notice that both algorithms perform the same logical operation, that is the final recall and provision are the same on all the datasets, hence not reported here.



Figure 14: Reparation vs. Broadcast meta-blocking. For each dataset we report two different strate ies for the **prune** functions, i.e., the weight- and cardinality-based prunin. This imes was taken on 10 nodes.

The results are reported in Figure 14: broadcast meta-blocking is faster t van rep: rtition meta-blocking from 4.9 to 12.7 times for WNP, and from 7.7 to 12.1 imes for CNP. To analyze the scalability of the algorithms, we report \dots Timer 15 their execution times in function of the number of nodes (from 1 to 10) on **freebase** (the largest dataset). In our setting, repartition m 'a-b ocking is not able to run with less than 7 nodes; whereas broadcast meta-blocking on a single node is 3 to 4 times faster than the execution time of the a partition meta-blocking on 10 nodes.



Figure 15: Scalability comparison: reparts on vs. broadcast meta-blocking on freebase.

We conclude that the broadc. * meta-blocking is always faster than the repartition meta-blocking.

5.5. Parallel-blast scalabili y

Finally, we assess the scal bill of parallel *Blast* by varying the number of nodes in the cluster (1–3, 5–7 and 10 nodes). For this experiment we employ **freebase**, which is the wise dataset to process due to the huge number of comparisons yielded by the blocking phase (2.23×10^{13} comparisons), and to its large number of "tributes (47,945 distinct attributes).

Figure 17 shows the realability of each blocking step, i.e.: Loose Schema Blocking (LSB, and this composed of Loose attribute-Match Induction in combination of Tok a Blocking), and Loose Schema Meta-Blocking (LS-MB). Figure 16 shows the spectup of each blocking step, which is sub-linear to the number of nodes in the cluster (i.e. 10x nodes, the overall speedup do not reach 5). For each the version of execution time from 1 to 3 nodes. The rescale of 4.2×10^{-1} speedup on 10 nodes of 4.2×10^{-1} .

T. e time : ad speedup reported so far only consider the blocking and metablocking, ' use of an ER process. In practice, all the comparisons generated through 'ny blocking process have to be compared by means of an *Entity Resc'ution A gorithm*, which is a binary function that takes as input two profiles and ' uses whether or not they are matching [36, 5]. Such a function is typically appendix, involving string similarity computations, calls to external



resources or even human intervention (i.e., crowdsourcing). Thus, the more the employed function is expensive, the more useful a good blocking (and metablocking) method is; in other words: the resources saved avoiding superfluous comparisons are proportional to the complexity of the *Entity* h solution Algorithm. Hence, we now compare Blast and WNP using a value (i.e., cheap) Entity Resolution Algorithm for showing that Blast significantly $\mathbf{1}$, funce the overall execution time of a complete ER process. We employ as Entity Resolution Algorithm the computation of the Jaccard Similarity of the two pr files involved in each comparison¹⁹.



Fi are d: Speedup of Blast on freebase.

 $^{19} {\rm In}$ a real-world scenario, a hreshold would be required to discriminate between matching and non-matching pa. 3.



Figure 17: Execution time of Blast on fre base.

Figure 18 shows the execution time of $Blost \, cm^2$ WNP in combination with the naïve *Entity Resolution Algorithm*²⁰ and by varying the number of nodes. We observe that the meta-blocking phase of *Llast* is slower than standard schema-agnostic WNP. This is not surprising, since *Blast* performs an additional step compared to WNP (i.e., *Loose & tribute-Match Induction*). Yet, the overall ER process employing *Blast* is s_{15} if cantly faster that employing WNP, since it retains much fewer comparison. ($3.80 \cdot 10^8$ of *Blast* vs. $2.17 \cdot 10^{10}$ of WNP). Please, recall that *Blast* and w NP, on freebase, achieve the same recall (Figure 11(g)).

 20 The average comparison th. on fre base is 0.05 ms.



Figure 18: Execution time of the complete ER process on **freebase**, varying the number of execution nodes in the cluster. The whole ER process is composed of a blocking phase, which generates candidate part that are compared through an Entity Resolution Algorithm. In (a), the blocking me hod employed is Token Blocking in combination with WNP meta-blocking. In (b), the blocking method employed is *Blast*.

5.6. LSH-based Loose Scheme Rlocking

This section aims at assussing t. e benefit of the LSH-based step. To do that, consider the worst case scen. "io: w' en Loose Schema Blocking (see Section 3.2) does not identify any similar attr.' ute, all the attributes are grouped in a unique all-encompassing cluster (the glue cluster [7]). In this scenario, the blocks generated combining Lc as a locking are identical to those generated with Token Blocking alore. On the other hand, if Loose Schema Blocking correctly groups some similar attributes, separating them from the glue cluster, the precision of the problem collection increases, while recall remains almost the same.

Ideally, the four of the similar attributes are correctly grouped, the higher the precision of the generated blocks is, without affecting the recall. Hence, to demonstrate the advantage of LSH-based Loose Schema Blocking, we perform a set of explaiments "disabling" the glue cluster and varying the threshold of LSH. This means that, without the glue cluster, all the attributes that are not indexed in a group of similar attributes are discarded, and so are the tokens of their values. If significant tokens are not employed as blocking key, the recall of the transitional blocks is negatively affected. So, varying the threshold of LSH changes the group of similar attributes. In fact, if two attributes are less similar²¹ than

Jaccard similarity, since we are employing min-hash.

the threshold, *Loose Schema Blocking* does not consider them as cardidate pair, and they cannot be indexed in the same group.

Figure 19 shows how LSH affects the final results of Blast comuned with Loose Schema Blocking in terms of recall on dbpedia (othe datasets yields analogous results). Table 5 reports the execution times of t^{1} excernment. We consider the recall of the block collection produced with Loose Sc. ma Blocking only, without considering the meta-blocking phase. Basic IIy, up to a threshold value of 0.35 (i.e., Jaccard similarity equals to 0.35), the recall is not affected (recall = 99.99%), meaning that $(almost^{22})$ all the matching r ofile pairs are successfully indexed in the block collection. The preasion is not reported, but for the points where recall = 99.99% is identical, i. is rot affected by the LSH threshold. For a threshold greater than 0.3, on the contrary, the techniques start failing to index some profile pairs, entain, 3 a degradation of the final result. In other words, for thresholds th. * exclu e too many attribute comparisons, Loose Schema Blocking fails to recognize similar attributes and produces an incomplete cluster of attributes. Neve. 'heless, even for a conservative threshold (e.g. 0.10), the execution of 'oose Schema Blocking, overall, is under 2h (instead of $\sim 12h$).



Figure 19: Recall with a. "r nt L' H configurations in combination with Loose Schema Blocking on dbpedia In the . "Ind number of rows and number of bands for LSH are in parenthesis, and ... the estimated threshold.

- [] LSF 0.10	$LSH_{0.22}$	$LSH_{0.32}$	$LSH_{0.41}$	$LSH_{0.55}$	$LSH_{0.64}$
12.5 h	$1.5 \ h$	$1.3 \ h$	1.2 h	0.9 h	0.7 h

Table 5: *Lose schema Blocking* run time varying the LSH threshold. The leftmost column report the xecution time of *Lose Schema Blocking* without employing LSH (i.e., computing ¹ - Jaccard similarity of all possible pair of attributes).

²Loose Schema Blocking (as any other blocking technique) may yield false negative, i.e., 1 airs of pr file that are not indexed in any block; for this reason the recall is not 100%.

	Blast	WNPOR	WNPAND	CNPOR	CNPAND	
recall(%)	74.7	78.3	68.3	84.4	78.7	
precision(%)	8.90	8.02	11.5	8.8	14.2	
F_1	0.1590	0.1448	0.1965	0.1608	0.2361	
	1k profile	s, Ground '	Truth: 300 1	matches		
(5 attri	butes - 2	clusters wit	th Loose Sch	ema Block	ing)	
		(a) ce	nsus			
	Blast	WNPOR	WNP_{AND}	CNPOR	CN AND	
recall(%)	82.1	90.3	81.2	66.9	46. ⁷	
precision(%)	84.0	53.8	69.4	65.7	52.4	
F_1	0.8302	0.6726	0.7377	0.6637	0.5 11	
1k profiles, Ground Truth: 17k matches						
(12 attr)	ibutes - 4	clusters wi	th Loose Sch	nema Blov'	ing)	
(b) cora						
	Blast	WNPOR	WNPAND	CNPOR	P _{AND}	
recall(%)	93.7	97.3	96.1	96.	94.9	
precision(%)	0.13	0.03	0.04	0.00	0.18	
F_1	0.0027	0.0005	0.0008	<u><u><u></u></u> 0015</u>	0.0036	
10k profiles, Ground Truth: 600 match.						
(106 attributes - 16 clusters with Loos Schrum Blocking)						
(c) cddb						
Table 6: 1 'r. EB results						
Table 0. E (1) the results.						

5.7. Dirty ER

Loose Schema Blocking is designed to identify similar attributes among data sources that have different schen, s (e.g. to identify which attributes refers to person names in the example ϵ . Figure 1). There is a particular class of Entity Resolution problems, c." d *dirty ER*, where single data source with known schema is condered, as outlined in [12] (see Section 2.1.1). In this scenario, there is inhere. I was need to perform loose attribute-match induction (or schema-alignment), because there is only a single source involved that has a unique schema. How wer, grouping similar attributes (if any) and extracting aggregate entropy is possible; thus, we modified *Loose Schema Blocking* to work with dirty ER see ection 2.1.1). For the meta-blocking phase, there is no need for changes.

To eval ate the performance of *Blast* we compared it against traditional meta-blocking techniques on 3 real-world benchmark datasets [1]. Both *Blast* and traditional meta-blocking are applied in combination with *Loose Schema Blocking*³.

 $^{2^{\}circ}$ raditional meta-blocking in combination with Token Blocking has always worse perforr ances, thus we do not report here the results. The execution times for these datasets are contract the order of milliseconds and *Loose Schema Blocking* does not significantly affect the total execution times.

ACCEPTED MANUSCRIPT

				#duplic les
	# profiles	#attributes	Amazon-TMD	760
IMDB	6.4k	12	Amazon-Rotten	5
Rotten	7.3k	16	Amazon-IMDB	2
Amazon	5.3k	6	IMDB-Rotten	871
TMD	10k	5	IMDB-TMD	5.
			TMD-Rotten	7.2

Table 7: Dataset characteristics: number of entity profiles, ε_{1d} number of *attribute* names. On the right side, the number of duplicates between ach data et.

Results

The characteristics of the datasets and the results are listed in Table 6. Besides recall and precision, we also consider F_1 -sco. which is the harmonic means of the two. This helps us to discuss the "omparise" of two methods that show significantly different values of both recall a. ⁴ r ecision. Blast achieves higher precision and F_1 -score than traditional W.^{-D}, and a slightly lower recall.

Overall, for dirty ER, *Blast* can e a effective blocking technique when the priority is to achieve high precision, without giving up a high level of recall (e.g., to save computational reservces performing ER in a cloud-computing environment).

5.8. Multiple Data Source

In this experiment we want to explore the multi-data source scenario [18], i.e. when the number of inv at datasets is greater than 2.

The datasets employ \sim' in t¹ is experiment have been collected from the Magellan repository [3¹], in particular we consider a collection of heterogeneous records gathered e and \sim from IMDB.com, RottenTomatoes.com, TMDmoviez.com and Amazon.com all about movies. These datasets have been used for evaluating ER algorit¹ ms in [5]. Considered singularly, none of these datasets contains duplicates; the \sim t¹ is ER problem can be formalized as a *Clean-Clean ER* problem (a.k.a. *Recored Linkage*) [14, 12] (see *Clean-Clean ER* in Section 2.1.1). Thus, *Ble t* and meta-blocking can be employed without any modification for this expendent \sim notice that if each dataset considered singularly could contain d¹ performed and the overall problem can be reduced to a *Dirty ER* problem (see f ection 1.1) on a single dataset that is the union of all the considered dataset is [12].

The quasaets characteristics are reported in Table 7. All the considered catasets are different schemas [5]. The ground truth has been generated using $t_1 \sim Mag'$ lan framework [5], the number of identified duplicates between each dataset are reported in Table 7.

ACCEPTED MANUSCRIPT



Figure 20 reports the achieved results. Blast obtains better results ' oth in term of recall and precision w.r.t the standard meta-blocking (Figure 2. (a-b)).

Figure 20: Recall and precision achieved by the onsuce d methods on the multisource datataset.

6. Related Work

Blocking techniques have been prime. It employed in Entity Resolution (ER) [37, 38, 14, 5, 39, 40, 41, 42, 43], and can be classified into two broad categories: *schema-based* (Suffix A prime [22], q-grams blocking [44], Canopy Clustering [45]), and *schema-agnostic* (Teken Blocking [7], Progressive ER [16, 46, 47, 48, 49, 50], and Attribute-match inauction [7, 9]).

Attribute-match induction—i mong the schema-agnostic techniques, Attribute Clustering (AC) [7]. and TY 'iMatch [9] try to extract statistics to define efficient blocking keys. AC reaction on the comparison of all possible pairs of attribute profiles of two class is to find the pairs of those most similar; this is a inefficient process, because are vist majority of comparisons are superfluous. Our LSH-based preprocessing step aims to address this specific issue. TYPiMatch tries to identify the reaction subtypes from generic attributes (e.g. description, info, etc.) frequent on generic dataset on the Web, and uses this information to select blocking key; but it cannot efficiently scale to large dataset.

Block maripul. 'on—In this paper, we tackled the problem of meta-blocking, i.e., how to rest ucture (*manipulate*) an existing blocking collection, for improving the q. lit of the overall ER process. The state-of-the-art, *unsupervised* and schema-egnosic neta-blocking has been presented in [12]. Blast was shown to outpe form them in Section 5. Supervised meta-blocking [51, 52] extends the blocking grap model by representing each edge as a vector of schema-agnostic features (). graph topological measures), and treats the problem of identif ing most promising edge as a *classification* problem; hence, a training set of lubeled d ta (matching/non-matching pairs) is required. Blast exploits the *loose* scheme information and does not require any training set (i.e., it is completely un equarvised).

Recently, in the context of multi data-source ER, Ranbadug et 1. [18] have proposed a blocking manipulation method for identifying entitle, whose profiles span among g data sources, where g is a user bounded parameter. In order to do that, given a block collection, the proposed me noc selects and combines (manipulate) blocks that are the most promising or finding profiles of q data sources that match together. The user can also spec fv a set F of data sources, and the final result is required to have matches that involves that set F of data sources. In [18], this task is called Multid tabase ecord linkage (MDRL). Formalizing MDRL by employing the blocking g. and nodel (Section 2.3): MDRL is the task of identifying the hypered set e^{f} the blocking graph that span among q nodes that belong to q distinct \cdots ces, and that are have high weight (remember that the weight in the blocking grand corresponds to the matching likelihood). Hence, the scope of MDRL is on bogonal to the scope of meta-blocking [12] (and thus Blast), which tries γ prune edges that correspond to not-promising comparisons. Furthermore, he examples MDRL solution [18] has been applied only in the context of structured . ta sources with well known schemas; while Blast does not require a p. reteined schema (since it relies on the loose schema information). Thus, the con. ination of the two methods is not trivial, but it is surely a future direction and two aim to explore, since the promising results achieved by Blast in the nulti source scenario of Section 5.8 (where the g and F parameters are h), considered).

Metadata exploitation—Thera is excellent related work in the semantic Web community [17, 53, 54, 55]. For instance, LIMES [53] (an ER approach for the Web of Data), and LOV [54] (a system attempting to standardize vocabularies) propose techniques to exploit a setadata, which may also be valuable to our problem, but are orthogonal to dur approach. In fact, *Blast* addresses the blocking problem based burgly on the attribute values, without considering the semantics of the schematical at all.

Entity Resolution w. b Mar Reduce-like Systems—Parallel and distributed versions of traditional (schemore-based) blocking techniques have been extensively studied in [56, 57]. A towim and Mehrota [58] have investigated how to generate candidate profile pairs on MapReduce-like systems in *pay-as-you-go* (i.e., progressive) fanion. Their proposed solution relies on the definition of schema-based blocking ke s. Finally, Efthymiou et al. [13] have proposed the *repartition meta blocking* algorithm to run graph-based meta-blocking methods on MapReduce. I Sections 4 and 5, we extensively compare it against our proposed broady user retarblocking algorithm.

Ar a_{J} of et a. [59] have proposed a novel schema-agnostic pruning strategy called *Globat Weighted Node Pruning* (GWPN) that combines a local threshold with a global one. The local threshold is computed for each profile as for the *NNF*, while the global one is computed as the average of all the edges vielts. This strategy aims to discard the edges with a low weight that connects on g profiles with a very low local threshold. Compared to traditional WNF, GWNP improves precision of 0.01%, while achieving the same recall, on D'spedia dataset [59]. Araújo et al. also discuss a Spark implementation for

their strategy, which is based on the MapReduce parallel meta-blocking proposed in [13], and suffers from the same limitations (see Section \wedge 2.2).

7. Conclusion and Future Work

In this paper we presented a holistic (meta-)blocking approac. Blast, able to automatically collect and exploit loose schema information 'i.e., statistics gathered directly from the data for approximately describing the data source schemas). We explained how this loose schema information in the extracted efficiently even from highly heterogeneous and volum more in tasets through an LSH-based step. We proposed a novel algorithm in efficiently running any meta-blocking technique on MapReduce-like Systems. S. Linally, we experimentally evaluated it on real-world datasets. The experimental results showed that: (i) Blast outperforms the existing state-of-the-an meta-blocking approaches in terms of quality of the results; (ii) our broadcan meta-blocking is always faster than the existing state-of-the-art when leveraging in distributed and parallel computation of MapReduce-like Systems.

Relevant research problem can be explored a. future directions: in the context of multi-data source ER, we aim to 'nvestigate how to combine our Loose-Schema Aware (meta-)blocking method with MDRL solution [18] (presented in Section 6). In the context of progressive FR (a.k.a. pay-as-you-go ER) [47], we aim to investigate how to exploit broad ast meta-blocking to yield progressive results, maximizing the recall on the basis of a limited resource budget (e.g., limited execution time, and/or computational resources). Finally, we are planning to combine our blocking technique for scaling to large data set advanced similarity functions that leverage on external knowledge bases, such as [60], with other MapReduce-like systems [61] and on real-world applications, such as the deduplication of web progressing [2].

- P. Christen, A surv of indexing techniques for scalable record linkage and deduplication, IEE. transactions on knowledge and data engineering 24 (9) (2012) 1557-1555.
- [2] X. L. Doné, D. Srivastava, Big data integration, Synthesis Lectures on Data Manager ant (1) (2015) 1–198.
- [3] S. Bergamaschi, D. Beneventano, F. Mandreoli, R. Martoglia, F. Guerra, M. Casir, L. Po, M. Vincini, G. Simonini, S. Zhu, et al., From data integration to big data integration, in: A Comprehensive Guide Through t'le Italian Database Research Over the Last 25 Years, Springer, 2018, pp. 43-59.
- ^[2] R. Baxter, P. Christen, T. Churches, et al., A comparison of fast blocking met. ods for record linkage, in: ACM SIGKDD, Vol. 3, Citeseer, 2003, pp. 25-^c 7.

- [5] P. Konda, S. Das, P. Suganthan GC, A. Doan, A. Ardalan, J. L. Ballard, H. Li, F. Panahi, H. Zhang, J. Naughton, et al., Magellan: "oward building entity matching management systems, Proceedings of "o VLDB Endowment 9 (12) (2016) 1197–1208.
- [6] J. Madhavan, S. R. Jeffery, S. Cohen, X. Dong, D. Ko, C. 'u, A. Halevy, Web-scale data integration: You can only afford to pay as jou go, in: Proceedings of CIDR, 2007, pp. 342–350.
- [7] G. Papadakis, E. Ioannou, T. Palpanas, C. Nieder, W. Mujdl, A blocking framework for entity resolution in highly heterogeneous performation spaces, IEEE Transactions on Knowledge and Data Engineering 25 (12) (2013) 2665-2682.
- [8] G. Papadakis, G. Koutrika, T. Palpanas, V. Nejdl, Meta-blocking: Taking entity resolution to the next level, IEEL Trans. Jons on Knowledge and Data Engineering 26 (8) (2014) 1946–1960.
- [9] Y. Ma, T. Tran, Typimatch: Type-specify unsupervised learning of keys and key values for heterogeneous to Lota integration, in: Proceedings of the sixth ACM international conference on Web search and data mining, ACM, 2013, pp. 325–334.
- [10] C. E. Shannon, A mathematical t. eory of communication, SIGMOBILE Mob. Comput. Commun. Re. 5 (1) (2001) 3-55. doi:10.1145/584091. 584093.
- [11] G. Simonini, S. Berga asch, H. Jagadish, Blast: a loosely schema-aware meta-blocking appro. ^h for e tity resolution, Proceedings of the VLDB Endowment 9 (12) [2016] ¹¹ 3–1184.
- [12] G. Papadakis, G. Pap stef natos, T. Palpanas, M. Koubarakis, Scaling entity resolution 'D large heterogeneous data with enhanced meta-blocking., in: EDBT, 2010 pp. 221–232.
- [13] V. Efthymⁱ G. Papadakis, G. Papastefanatos, K. Stefanidis, T. Palpanas, Pr alle meta-blocking for scaling entity resolution over big heterogeneous da a Information Systems 65 (2017) 137–157.
- [14] P. C¹ risten, Data Matching Concepts and Techniques for Record Linkage, E. Cy R solution, and Duplicate Detection, Data-Centric Systems and A Plication, Springer, 2012. doi:10.1007/978-3-642-31164-2.
- [15] \setminus Christ phides, V. Efthymiou, K. Stefanidis, Entity resolution in the web of a. , Synthesis Lectures on the Semantic Web 5 (3) (2015) 1–122.
- [6] G. S. monini, G. Papadakis, T. Palpanas, S. Bergamaschi, Schema-agnostic progressive entity resolution, IEEE Trans. Knowl. Data Eng. (2018)doi: 10.1109/TKDE.2018.2852763.

ACCEPTED MANUSCRIPT

- [17] P. Shvaiko, J. Euzenat, Ontology matching: state of the art and inture challenges, IEEE Transactions on knowledge and data engineering 25 (1) (2013) 158-176.
- [18] T. Ranbaduge, D. Vatsalan, P. Christen, A scalable and efficient abgroup blocking scheme for multidatabase record linkage, in: Pac.² -Asia Conference on Knowledge Discovery and Data Mining, Springer, 2016, pp. 15–27.
- [19] A. Z. Broder, On the resemblance and containment of documents, in: Compression and complexity of sequences 1997. proceedings, LeEE, 1997, pp. 21–29.
- [20] J. Leskovec, A. Rajaraman, J. D. Ullman, Mining A massive datasets, Cambridge university press, 2014.
- [21] T. M. Cover, J. A. Thomas, Elements c⁺ into matter into theory, John Wiley & Sons, 2012.
- [22] T. De Vries, H. Ke, S. Chawla, P. Chr. en, Robust record linkage blocking using suffix arrays, in: Proceedings of the 18th ACM conference on Information and knowledge manage perts, ACM, 2009, pp. 305–314.
- [23] A. Agresti, M. Kateri, Categoric in ¹ata nalysis, in: International encyclopedia of statistical science, Spring, v, 2011, pp. 206–208.
- [24] M. Zaharia, M. Chowdhury, T. Das, A. Dave, J. Ma, M. McCauly, M. J. Franklin, S. Shenker, I. Stoica, Resilient distributed datasets: A faulttolerant abstraction for memory cluster computing, in: Presented as part of the 9th USE^T IX Syn posium on Networked Systems Design and Implementation (NSDL 1⁽²⁾). USENIX, San Jose, CA, 2012, pp. 15–28.
- [25] J. Dean, S. Ghe law i, Mapreduce: simplified data processing on large clusters, Communic. ions of the ACM 51 (1) (2008) 107–113.
- [26] [link]. URL https://spark.apache.org/docs/2.1.0/programming-guide. html#shu fle operations
- [27] A. S. Das, M. Datar, A. Garg, S. Rajaram, Google news personalization: scalable or line collaborative filtering, in: Proceedings of the 16th international conference on World Wide Web, ACM, 2007, pp. 271–280.
- [28] S. Blanes, J. M. Patel, V. Ercegovac, J. Rao, E. J. Shekita, Y. Tian, A com-1 arison o join algorithms for log processing in mapreduce, in: Proceedings of the 2/10 ACM SIGMOD International Conference on Management of data, ACM, 2010, pp. 975–986.
- [2] [link]. URL http://stravanni.github.io/blast/

ACCEPTED MANUSCRIPT

- [30] H. Köpcke, A. Thor, E. Rahm, Evaluation of entity resolution approaches on real-world match problems, Proceedings of the VLDB Endowmed + 3 (1-2) (2010) 484-493.
- [31] S. Das, A. Doan, P. S. G. C., C. Gokhale, P. Konde i'e magellan data repository, https://sites.google.com/site/nhaidgroup/ projects/data.
- [32] A. Harth, Billion triples challenge data set (2012).
- [33] D. Hand, P. Christen, A note on using the f-me sure ^r evaluating record linkage algorithms, Statistics and Computing 2 (2) (20 8) 539-547.
- [34] M. Ebraheem, S. Thirumuruganathan, S. Joty, N. Ouzzani, N. Tang, Distributed representations of tuples for entity resolution, Proceedings of the VLDB Endowment 11 (11) (2018) 1454–1467.
- [35] S. Mudgal, H. Li, T. Rekatsinas, A. Doon, V. rk, G. Krishnan, R. Deep, E. Arcaute, V. Raghavendra, Deep leaning for entity matching: A design space exploration, in: Proceedings of the 2013 International Conference on Management of Data, ACM, 2018, pp. 19-54.
- [36] O. Benjelloun, H. Garcia-Molina, D. Menestrina, Q. Su, S. E. Whang, J. Widom, Swoosh: a generic approach to entity resolution, The VLDB Journal—The International Commun. Market New York Large Data Bases 18 (1) (2009) 255-276.
- [37] H. Köpcke, E. Rahm, Frame orks for entity matching: A comparison, Data & Knowledge Engine ring 69 (2) (2010) 197–210.
- [38] F. Naumann, M. H rschel, A. Introduction to Duplicate Detection, Synthesis Lectures on Γ ata 'lang gement, Morgan & Claypool Publishers, 2010. doi:10.2200/S202 FD1' 01Y201003DTM003.
- [39] M. Stonebral er, D. Bruckner, I. F. Ilyas, G. Beskales, M. Cherniack, S. B. Zdonik, A. Pagan, S. Xu, Data curation at scale: The data tamer system, in: CIDR .01: Sixth Biennial Conference on Innovative Data Systems Research, A. lor ar, CA, USA, January 6-9, 2013, Online Proceedings, 2013.
- [40] G. Pepade'xis, L. Tsekouras, E. Thanos, G. Giannakopoulos, T. Palpanas, M. L'oul arak s, The return of jedai: End-to-end entity resolution for structured and semi-structured data, PVLDB 11 (12) (2018) 1950–1953. cbi:10.'4778/3229863.3236232.
- [41] V. Efthy miou, G. Papadakis, K. Stefanidis, V. Christophides, Simplifying entity resolution on web data with schema-agnostic, non-iterative matching, in: 5 lth IEEE International Conference on Data Engineering, ICDE 2018, Par.s, France, April 16-19, 2018, 2018, pp. 1296–1299.
 URL https://doi.org/10.1109/ICDE.2018.00134

- [42] A. D. Sarma, A. Jain, A. Machanavajjhala, P. Bohannon, C. LOCK: an automatic blocking mechanism for large-scale de-duplication tasks, CoRR abs/1111.3689. arXiv:1111.3689.
 URL http://arxiv.org/abs/1111.3689
- [43] U. Draisbach, F. Naumann, A generalization of blocking ind windowing algorithms for duplicate detection, in: 2011 International Conference on Data and Knowledge Engineering, ICDKE 2011, Mi ano, Italy, September 6, 2011, 2011, pp. 18-24. doi:10.1109/ICDKE.2011.30539^c J.
- [44] L. Gravano, P. G. Ipeirotis, H. V. Jagadish, N. Loudes, C. Muthukrishnan, D. Srivastava, Approximate string joins in a database (*r* most) for free, in: VLDB 2001, Proceedings of 27th International Conference on Very Large Data Bases, September 11-14, 2001, Roma, Italy, 2–01, pp. 491–500.
- [45] A. McCallum, K. Nigam, L. H. Unger Endert clustering of highdimensional data sets with application to reference on Knowledge discovery and data mining, Boston, MA, CCA, August 20-23, 2000, 2000, pp. 169–178. doi:10.1145/347090 344 LCC
- [46] G. Simonini, G. Papadakis, T. J. bana S. Bergamaschi, Schema-agnostic progressive entity resolution, in: 34. IEEE International Conference on Data Engineering, ICDE 20¹⁰ Para France, April 16-19, 2018, 2018, pp. 53-64. doi:10.1109/ICDE.201² 00015.
- [47] S. E. Whang, D. Marmong, H. Garcia-Molina, Pay-as-you-go entity resolution, IEEE Trans. Knowl. Data Eng. 25 (5) (2013) 1111–1124. doi: 10.1109/TKDE.2012.4
- [48] T. Papenbrock, A. Helse, F. Naumann, Progressive duplicate detection, IEEE Trans. Knowl Dat. Eng. 27 (5) (2015) 1316–1329. doi:10.1109/ TKDE.2014.23'9666.
- [49] D. Firmani, B. Saha, D. Srivastava, Online entity resolution using an oracle, PVLDB 9 (a) (2016) 384-395. doi:10.14778/2876473.2876474. URL http://www.vldb.org/pvldb/vol9/p384-firmani.pdf
- [50] D. Fir nani, S. Calhotra, B. Saha, D. Srivastava, Robust entity resolution using a cowderacle, IEEE Data Eng. Bull. 41 (2) (2018) 91-103. URL http://sites.computer.org/debull/A18june/p91.pdf
- [51] G. Papacakis, G. Papastefanatos, G. Koutrika, Supervised meta-blocking, *P*⁷LDB (14) (2014) 1929–1940. doi:10.14778/2733085.2733098.
- [2] G. Cul Bianco, M. A. Gonçalves, D. Duarte, Bloss: Effective meta-blocking with almost no effort, Information Systems 75 (2018) 75–89.

- [53] A. N. Ngomo, S. Auer, LIMES A time-efficient approach for large-scale link discovery on the web of data, in: IJCAI 2011, Proceeding, of the 22nd International Joint Conference on Artificial Intelligence, Procedona, Catalonia, Spain, July 16-22, 2011, 2011, pp. 2312–2317 doi:10.5591/ 978-1-57735-516-8/IJCAI11-385.
- [54] P. Vandenbussche, B. Vatant, Linked open vocabu'arres, Er.CIM News 2014 (96).
- [55] S. Bergamaschi, D. Ferrari, F. Guerra, G. Simonini, Y. Mograkis, Providing insight into data source topics, J. Data Seminatics of (1) (2016) 211–228. doi:10.1007/s13740-016-0063-6.
- [56] L. Kolb, A. Thor, E. Rahm, Dedoop: Efficient dec. plication with hadoop, PVLDB 5 (12) (2012) 1878–1881. doi:10. 4778/2 67502.2367527.
- [57] S. Das, P. S. G. C., A. Doan, J. F. Naugl on, G. Krishnan, R. Deep, E. Arcaute, V. Raghavendra, Y. Park Taken, Scaling up hands-off crowdsourced entity matching to build cloud services, in: Proceedings of the 2017 ACM International Conference or Intergement of Data, SIGMOD Conference 2017, Chicago, IL, USA, M. v 14-19, 2017, 2017, pp. 1431–1446. doi:10.1145/3035918.303596
- [58] Y. Altowim, S. Mehrotra, Parallel progressive approach to entity resolution using mapreduce, in: 33rd ILTE international Conference on Data Engineering, ICDE 2017, San Diego, CA, USA, April 19-22, 2017, 2017, pp. 909–920. doi:10.1109/TCF.2017.139.
- [59] T. B. Araújo, C. E. Pires, J. P. da Nóbrega, Spark-based streamlined metablocking, in: Computer 3 and Communications (ISCC), 2017 IEEE Symposium on, IF EE. 2017, pp. 844–850.
- [60] F. Benedetti, J. Bengeratano, S. Bergamaschi, G. Simonini, Computing inter-document crimilarity with context semantic analysis, Inf. Syst. 80 (2019) 136-147. doi: 10.1016/j.is.2018.02.009.
- [61] S. Bergar asch, L. Gagliardelli, G. Simonini, S. Zhu, Bigbench workload executed b, ' sing apache flink, Procedia Manufacturing 11 (2017) 695–702.
- [62] F. G erre, G. Simonini, M. Vincini, Supporting image search with tag clouds. A p eliminary approach, Adv. in MM 2015 (2015) 439020:1– 4? 20:10. .oi:10.1155/2015/439020.

Highlights

- An unsupervised graph-based meta-blocking approach (called Blast) able to leverage this loose schema information;
- an LSH-based attribute-match induction technique for efficiently scale courge datasets with a high number of attributes;
- an algorithm to efficiently run Blast (and any other graph-based 'nc.a-blocking method) on MapReduce-like systems, to take full advantage of a parallel an 'distributed computation;
- the evaluation of our approach on seven real-world dataset ;, showing how Blast outperforms the state-of-the-art meta-blocking methods.